Kimberly Caswell

Vice President and General Counsel, Southeast Legal Department

RECEIVED-1750

U2 DEC 19 AM 10: 22

COMMISSION CLERK



FLTC0007 201 North Franklin Street (33602) Post Office Box 110 Tampa, Florida 33601-0110

Phone 813 483-2606 Fax 813 204-8870 kimberly.caswell@verizon.com

December 19, 2002

Ms. Blanca S. Bayo, Director Division of Commission Clerk and Administrative Services Florida Public Service Commission 2540 Shumard Oak Boulevard Tallahassee, FL 32399-0850

Re:

Docket No. 981834-TP

Petition of Competitive Carriers for Commission Action to Support Local Competition in BellSouth Telecommunications Inc.'s Service Territory

Docket No. 990321-TP

Petition of ACI Corp. d/b/a Accelerated Connections, Inc. for generic investigation to ensure that BellSouth Telecommunications, Inc., Sprint-Florida, Incorporated, and GTE Florida Incorporated comply with obligation to provide alternative local exchange carriers with flexible, timely, and cost-efficient physical collocation

Dear Ms. Bayo:

Please find enclosed an original and fifteen copies of the Direct Testimony of John Ries on behalf of Verizon Florida Inc. for filing in the above matters. Service has been made as indicated on the Certificate of Service. If there are any questions regarding this filing, please contact me at 813-483-2617.

Sincerely,

Sincerely,

Sincerely,

Sincerely,

Constituted of the second of the secon

02 DEC 19 MIC: 12

DISTRIBUTION CENTER

DOCUMENT AT MACH-BATE

13818 DEC 198

FPSC-COMMISSION CLERK

RECEIVED & FILED

FPSC-BUREAU OF RECORDS

CERTIFICATE OF SERVICE

I HEREBY CERTIFY that copies of the Direct Testimony of John Ries on behalf of Verizon Florida Inc. in Docket Nos. 981834-TP and 990321-TP were sent via U. S. mail on December 19, 2002 to the parties on the attached list.

Kimberly Caswell

Staff Counsel Florida Public Service Commission 2540 Shumard Oak Boulevard Tallahassee, FL 32399-0850

Nancy Sims BellSouth Telecomm. Inc. 150 S. Monroe Street, Suite 400 Tallahassee, FL 32301-1556 Rodney L. Joyce Shook Hardy & Bacon LLP 600 14th St NW, Suite 800 Washington, DC 20005-2004

Richard D. Melson Gabriel E. Nieto Hopping Law Firm 123 S. Calhoun Street Tallahassee, FL 32314

Virginia C. Tate/Lisa A. Riley AT&T 1200 Peachtree Street N.E. Suite 8066 Atlanta, GA 30309-3523 Norton Cutler Development Specialists Inc. c/o Steve Victor 70 West Madison Street Suite 2300 Chicago, IL 60602-4250

Peter M. Dunbar Barbara Auger Pennington Law Firm 215 S. Monroe St., 2nd Floor Tallahassee, FL 32301 Time Warner Telecom 2301 Lucien Way, Suite 300 Maitland, FL 32751 Nanette S. Edwards ITC^DeltaCom 4092 S. Memorial Parkway Huntsville, AL 35802-4343

Kenneth Hoffman Rutledge Law Firm 215 S. Monroe St., Suite 420 Tallahassee, FL 32302 Andrew Isar Telecomm. Resellers Assn. c/o Miller Isar, Inc. 7901 Skansie Ave., Suite 240 Gig Harbor, WA 98335 C. Pellegrini/Patrick Wiggins Katz Kutter Law Firm 106 E. College Avenue 12th Floor Tallahassee, FL 32301

Terry Monroe/Genevieve Morelli CompTel 1900 M Street N.W. Suite 800 Washington, DC 20036

Michael A. Gross Florida Cable Telecomm. Assn. 246 E. 6th Avenue, Suite 100 Tallahassee, FL 32303 Vicki Kaufman/Joe McGlothlin McWhirter Law Firm 117 S. Gadsden Street Tallahassee, FL 32301

David Tobin Fla. Public Telecomm. Assn. c/o Tobin & Reyes 7251 W. Palmetto Park Road #205 Boca Raton, FL 33433-3487 John D. McLaughlin, Jr. KMC Telecom Inc. 1755 North Brown Road Lawrenceville, GA 30043-8119 Deborah Eversole General Counsel Kentucky Public Service Comm. P. O. Box 615 Frankfort, KY 40602

Donna McNulty MCI WorldCom 1203 Governors Square Blvd. Suite 201 Tallahassee, FL 32301-2960 Floyd R. Self/Tracy Hatch Messer Law Firm P. O. Box 1876 Tallahassee, FL 32302

David Woodsmall Mpower Comm. Corp. 175 Sully's Trail, Suite 300 Pittsford, NY 14534-4558 Mark E. Buechele Supra Telecommunications 2620 SW 27th Avenue Miami, FL 33133

Laura L. Gallagher MediaOne Florida Tele. 101 E. College Avenue Suite 302 Tallahassee, FL 32301 Don Sussman Network Access Solutions Corp. Three Dulles Tech Center 13650 Dulles Technology Drive Herndon, VA 20171-4602

William H. Weber Covad Communications Co. 1230 Peachtree Street N.E. 19th Floor Atlanta, GA 30309-3574 S. Masterson/C. Rehwinkel Sprint-Florida Incorporated 1313 Blairstone Road MC FLTLHO0107 Tallahassee, FL 32301 Carolyn Marek Time Warner Telecom 233 Bramerton Court Franklin, TN 37069

Bettye Willis
ALLTEL Communications
Services Inc.
One Allied Drive
Little Rock, AR 72203

J. Jeffry Wahlen Ausley & McMullen 227 S. Calhoun Street Tallahassee, FL 32302 Anu Seam U.S. Department of Justice Telecom Task Force Antitrust Division 1401 H Street N.W., Suite 8000 Washington, DC 20530

Anita L. Fourcard Lockheed Martin IMS Comm. Industry Services 1200 K Street, N.W. Washington, DC 20005 Brent McMahan Network Telephone Corporation 815 South Palafox Street Pensacola, FL 32501

BEFORE THE FLORIDA PUBLIC SERVICE COMMISSION

In re: Petition of Competitive Carriers for Commission action to support local competition in BellSouth Telecommunications Inc.'s service territory

Docket No. 981834-TP

In re: Petition of ACI Corp. d/b/a Accelerated Connections, Inc. for generic investigation to ensure that BellSouth Telecommunications, Inc., Sprint-Florida, Incorporated, and GTE Florida Incorporated comply with obligation to provide alternative local exchange carriers with flexible, timely, and cost-efficient physical collocation.

Docket No. 990321-TP

JOHN RIES

ON BEHALF OF

VERIZON FLORIDA INC.

DECEMBER 19, 2002

DOCUMENT NUMBER DATE

FPSC-COMMISSION CLERK

1 Q. PLEASE STATE YOUR NAME AND BUSINESS ADDRESS.

2 A. My name is John Ries. My business address is 600 Hidden Ridge,

3 Irving, Texas 75038.

5 Q. BY WHOM ARE YOU EMPLOYED AND IN WHAT CAPACITY?

6 A. I am employed by Verizon Communications as a Senior Product

Manager - Collocation. In this proceeding, I am representing Verizon

8 Florida Inc.

Α.

Q. PLEASE DESCRIBE YOUR EDUCATION AND WORK EXPERIENCE.

I graduated from the University of Missouri - Columbia in 1982 with a Bachelor of Arts degree in Mathematics and Statistics. My employment with GTE (now Verizon) commenced in May 1982 in the Network Planning Department. I held several positions during my first six years with Network Planning. My responsibilities included capital budgeting, capital portfolio management, implementation of enhanced support products for Network Planning, and coordination of technical responses for business customer requests. In 1988, I moved into the Business Pricing group and remained there for four years. My responsibilities there included pricing new network services for tariff offerings, as well as pricing individual case applications.

In December 1992, I became the Product Manager for Expanded Interconnection Services. My responsibilities included coordinating GTE's response to the FCC's Docket 91-141 Order on Special Access

1		and Switched Transport Interconnection, a task which required
2		organizing diverse resources within GTE to determine how the
3		Company would offer physical and virtual collocation.
4		
5		In January 1998, I became Program Manager, Access Services. I was
6		involved in analyzing competitive information relating to GTE's Network
7		Services, as well as contract negotiations with major interexchange
8		carriers and competitive local exchange carriers.
9		
10		In January 2000, I moved into my current position, Senior Product
11		Manager - Collocation. Over the last two years, I have been a policy
12		witness on collocation issues and have negotiated collocation
13		interconnection agreements.
14		
15	Q.	HAVE YOU PREVIOUSLY TESTIFIED BEFORE ANY STATE
16		REGULATORY COMMISSIONS?
17	A.	Yes, I have testified on collocation issues in California, Florida, Hawaii
18		Illinois, Missouri, Nebraska, New Hampshire, New Mexico, North
19		Carolina, Rhode Island, Texas, Vermont, Washington, and Wisconsin.
20		
21	Q.	WHAT IS THE PURPOSE OF YOUR TESTIMONY?
22	A.	My testimony discusses Verizon's collocation policies, practices, and
23		terms and conditions. I cover all issues designated for resolution in this
24		docket, except for issues 9 and 10, concerning collocation rates
25		Varizon witness Ellis will address those issues

-1

2

3

4

5

6

7

8

A.

HOW DOES VERIZON PROVIDE COLLOCATION IN FLORIDA? Q.

Verizon provides collocation under a tariff, which took effect on January 14, 2000. Prior to that, collocation was provided through negotiated interconnection agreements. Verizon's tariff addresses most of the issues in this proceeding, and my testimony is consistent with the tariff provisions. The tariff is attached as Exhibit JR-1. Various sections of the tariff are referenced throughout my testimony.

9

13

14

15

16

17

18

19

20

21

Α.

WHEN SHOULD AN ALEC BE REQUIRED TO REMIT PAYMENT 10 Q. 11

FOR NON-RECURRING CHARGES FOR COLLOCATION SPACE?

12 (ISSUE 1A)

> Once Verizon confirms that it can satisfy the ALEC's collocation request, the ALEC should be required to remit 50% of the non-recurring charges associated with the collocation arrangement. Verizon's confirmation includes a price quote and project implementation timeline. Verizon will begin to prepare the space upon receipt of the initial 50% payment, which establishes the ALEC's commitment to proceed with the requested collocation and covers a portion of Verizon's up-front costs to prepare the collocation space. The remaining 50% of the non-recurring charges are billed to the ALEC at the time space is turned over to the ALEC.

23

24

25

22

WHEN SHOULD BILLING OF MONTHLY RECURRING CHARGES Q. **BEGIN? (ISSUE 1B)**

1 A. Billing of monthly recurring charges should begin in the next billing cycle
2 after the collocation space is turned over to the ALEC.

Q.

Α.

WHAT CANCELLATION CHARGES SHOULD APPLY IF AN ALEC CANCELS ITS REQUEST FOR COLLOCATION SPACE? (ISSUE 1C)

Verizon does not assess "cancellation charges." Rather, depending on when the ALEC cancels its request, Verizon would charge the ALEC for the costs Verizon incurred in responding to the request. If the ALEC cancels its request for collocation space when construction is in progress and prior to acceptance of the space, Verizon would keep the engineering/augment fees submitted with the application and would assess any other non-recurring charges necessary to cover Verizon's costs incurred in providing the collocation arrangement. If the ALEC cancels the request after the collocation arrangement has been completed, the ALEC would also have to pay the monthly recurring charges, unless the ALEC provided written notice of the cancellation 30 days prior to the scheduled completion date.

A.

Q. WHAT SECTIONS OF VERIZON'S TARIFF ADDRESS ISSUE ONE?

The following sections address the various aspects of bill payment and cancellation: Application Form/Fee (19.3.1.B), Notification of Acceptance/Rejection (19.3.1.C), Space Availability (19.3.2), Price Quote (19.3.3), Planning and Coordination (19.4.1), Cancellation and Acceptance Delays (19.10.3), and Billing and Payment (19.14.2).

1 Q. SHOULD AN ALEC BE REQUIRED TO JUSTIFY ITS SPACE
2 RESERVATION NEEDS TO THE ILEC WHEN AN ILEC IS FORCED
3 TO CONSIDER A BUILDING ADDITION TO ACCOMMODATE
4 FUTURE SPACE REQUIREMENTS? (ISSUE 2A)

It is first important to clarify that an ILEC cannot be forced to consider a building addition to accommodate either existing collocation requests or future ALEC collocation demand. But when an ALEC has reserved space in an office that is at or near exhaust and another party (either another ALEC or the ILEC) needs the space, the ALEC should be required to justify the space reservation by showing the intended purpose of the reserved space and the forecasted year of use. These are the same showings Verizon is required to make when it requests a waiver of collocation requirements due to space exhaustion. (See Verizon Collocation Tariff at sec. 19.5.1.) Applying them to the ALEC in this analogous context satisfies the same policy of efficiently using space to meet existing needs.

Α.

A.

18 Q. UNDER WHAT CONDITIONS SHOULD AN ILEC BE ALLOWED TO 19 RECLAIM UNUSED COLLOCATION SPACE? (ISSUE 2B)

Under Verizon's tariff, the ALEC must begin installing collocation equipment (that is, equipment necessary for interconnection or access to unbundled network elements) within a reasonable period of time, not to exceed six months, from the date the collocation arrangement is accepted. If the ALEC does not utilize its space within that time, Verizon may reclaim that unused space. This practice assures timely and

efficient use of limited space and prevents an ALEC from warehousing space in order to keep other competitors out of a market. These are the same objectives the Commission emphasized in its May 2000 Collocation Order. (See, e.g., Order No. PSC-00-0941-FOF-TP, at 54-55.)

Α.

7 Q. WHAT OBLIGATIONS, IF ANY, SHOULD BE PLACED ON THE ALEC 8 THAT CONTRACTED FOR THE SPACE? (ISSUE 2C)

Verizon's tariff comprehensively describes the obligations of the ALEC contracting for collocation space. With regard to general obligations on use of the space, as noted, the ALEC must begin to use its space within six months. In addition, if there is not enough space to satisfy existing collocation requests, an ALEC may not house obsolete or unused equipment within its space and it must document its plans for use of reserved space. Again, Verizon has the same obligations in a space exhaust situation. (See Verizon Collocation Tariff § 19.5.1.) In order to ensure the most efficient use of space, these obligations should apply equally to all parties using space in an office.

Q. WHAT OBLIGATIONS, IF ANY, SHOULD BE PLACED ON THE ILEC? (ISSUE 2D)

A. Verizon's tariff sets forth both the ILEC's and ALEC's obligations relative to collocation arrangements. With regard to the ILEC's use of space, Verizon must justify and document its existing use of space and its future needs before it may receive a waiver of collocation requirements

1		at a particular site. (See Verizon Collocation Tariff § 19.5.1)
2		
3	Q.	WHAT SECTIONS OF THE VERIZON TARIFF ADDRESS ISSUE
4		TWO?
5	A.	Along with the Space Availability section (19.5.1), the following sections
6		of Verizon's tariff also relate to this issue: Minimum/Maximum/Additional
7		space (19.5.2), Use of Space (19.5.3), Reservation of Space (19.5.4)
8		and Reclamation (19.5.6).
9		
10	Q.	SHOULD AN ALEC HAVE THE OPTION TO TRANSFER ACCEPTED
11		COLLOCATION SPACE TO ANOTHER ALEC? IF SO, WHAT ARE
12		THE RESPONSIBILITIES OF THE ILEC AND ALECS? (ISSUE3)
13	A.	No. While an ALEC may sublease its collocation space to another party
14		(pursuant to section 19.2.3 of Verizon's tariff), it may not transfer the
15		entire space to another ALEC once the contracting ALEC decides to
16		leave it. Verizon is responsible for the management and operation of its
17		central offices, including collocation space. A transfer of space to a third
18		party, without Verizon's input or knowledge, would undermine Verizon's
19		ability to control and maintain its premises.
20		
21		Allowing ALECs to transfer space to each other would also be directly
22		contrary to the Commission's November 2000 ruling on post-waiver
23		space availability. As the Commission knows, ILECs may receive FCC
24		waivers of physical collocation requirements where space is exhausted.
25		This Commission requires ILECs to keep waiting lists of ALECs that

have been denied physical collocation for lack of space in a particular Under this system, ALECs must be listed in the order their requests are received, so that if space later becomes available, the first ALEC application received must be given the first opportunity to take the (November 2000 Collocation Order at 20-21.) Allowing an ALEC to transfer space directly to another ALEC would circumvent the Commission's mandatory waiting list procedure. Without the ILEC's involvement, the ALEC could transfer the newly available space to any other ALEC, regardless of its position on the ILEC's waiting list. The ALEC could simply give the space to the highest bidder or use any other criterion it wished to allocate the space. This is exactly the kind of arbitrary and unfair result the Commission sought to prevent in its November 2000 Collocation Order. If the Commission considers allowing direct ALEC-to-ALEC transfers of space, it will necessarily have to change its post-waiver space allocation policies.

16

20

21

22

23

24

25

Α.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

17 Q. SHOULD THE ILEC BE REQUIRED TO PROVIDE COPPER 18 ENTRANCE FACILITIES WITHIN THE CONTEXT OF A 19 COLLOCATION INSIDE THE CENTRAL OFFICE? (ISSUE 4)

No. Verizon will allow the ALEC to bring fiber optic facilities into the ILEC premise, but it should not be forced to provide copper facilities, which take up significantly more space within the ILEC manhole and conduit system than fiber facilities. Moreover, fiber facilities can handle high volumes of traffic at higher bandwidth over a single fiber pair. Increasing conduit space to accommodate additional copper cable is a

labor-intensive and costly exercise. Verizon's tariff sections relating to this issue are 19.4.3.D and 19.4.3.E.

3

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

Α.

1

2

Q. SHOULD AN ILEC BE REQUIRED TO OFFER, AT A MINIMUM, POWER IN STANDARDIZED INCREMENTS? IF SO, WHAT SHOULD THE STANDARDIZED POWER INCREMENTS BE? (ISSUE 5)

Verizon does not oppose offering power in standardized increments, as long as ALECs order and maintain a specified minimum amperage. this regard. Verizon offers DC Power in per-amp increments, but requires a minimum of ten (10) amps for each ALEC arrangement. Ten amps is a reasonable minimum because a functioning collocation arrangement will require at least 10 amps of power. The minimum requirement is necessary for Verizon to have an opportunity to recover its costs, because while ALECs may require different quantities of power, power is not provisioned or grown at a single amp increment. Power rates must cover not only the costs specific to the particular arrangement (such as extending cabling from Verizon's power plant to a battery distribution fuse bay (BDFB); provisioning fusing, and extending cable to the collocation arrangement), but the ongoing costs of maintaining and investing in power plant infrastructure adequate to satisfy collocators' needs. While Verizon agrees to sell power on a peramp basis, the minimum amperage requirement is consistent with the bulk nature of the costs of provisioning power and minimizes the threat of stranded investment.

25

Note that while fusing and power cabling material costs vary directly with the number of amps provisioned, the labor to install different size cables is closely tied to the distance and number of cables to be placed, rather than on amperage quantities. Setting non-recurring charges on a percable installation basis and monthly recurring charges on a single amp increment with a 10 amp minimum attains the best balance.

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

Q.

A.

1

2

3

4

5

6

ON AMPS USED OR FUSED CAPACITY. CAN YOU EXPLAIN THE DIFFERENCE BETWEEN A LOAD AMP AND A FUSED AMP?

Telecommunications equipment that uses DC power has a manufacturer's power drain specification associated with it. instance, a particular piece of switching or transmission equipment may require 20 amps of DC power to function. This drain on the ILEC power plant is also referred to as the power "load." Today's digital technology consistently draws a steady load amperage from the power plant, but there may be instances when it will draw more or less amperage. The power cabling feed from the ILEC power plant is fused to accommodate a spike in the power drain from the equipment before the fuse will blow and take that cable feed out of service. Typically, the fuse is engineered to 1.25 or 1.50 times the load. In the above example, Verizon would fuse the power feed at 30 amps for the 20-amp load. Therefore, the ALEC's equipment could draw up as much as 30 amps of power from the ILEC without blowing a fuse. However, based on the manufacturer's specifications, the equipment would be expected to consistently draw 20

amps of power while in service. The usual debate underlying Issue 6A is whether the ALEC should pay for the fuse power of 30 amps or the load power of 20 amps.

Α.

Q. HOW DOES THE ALEC ORDER POWER FROM VERIZON?

The ALEC must request the number of power cables to be provisioned to the collocation arrangement, along with the power load and fuse quantity of each individual cable. Verizon allows the ALEC to request fusing up to two and one-half times the requested load on each power feed. This approach allows the ALEC to have full redundancy on its power feeds (to allow for failure on one feed), while only paying for the power load that was ordered for its equipment. For instance, if an ALEC required 20 amps of power for its collocation arrangement, a probable configuration would request 10 amps of power on the A cable feed and 10 amps of power on the B cable feed. This would provide the ALEC with the power load required for its equipment. The ALEC could also request that Verizon fuse each cable feed at 25 amps. Therefore, if one power lead were to fail, the other feed would keep the equipment inservice. In this example, Verizon would charge the ALEC for 20 amps of power.

Q.

Α.

SHOULD AN ILEC'S PER AMPERE (AMP) RATE FOR THE PROVISIONING OF DC POWER TO AN ALEC'S COLLOCATION SPACE APPLY TO AMPS USED OR FUSED CAPACITY? (ISSUE 6A) Neither used amps or fused amps should apply. Consistent with

Verizon's tariff, the per-amp rate should be based on what the ALEC orders. When an ALEC orders power, it specifies the load (the typical drain, based on manufacturer's specifications) and the fused capacity (how much of a power spike the fuses should accommodate). Verizon charges for power on a per-load-amp basis, rather than charging for the total fused amps or measuring a used amount. However, because Verizon fuses each power feed based on the ALEC's application if a ALEC abuses this pricing structure and consistently draws more power than it requested, Verizon should continue to have the ability to audit power usage and impose penalties for any abuses. As shown in the previous example, while the ALEC was only charged for 20 amps of power, the ALEC had a total of 50 amps of power being delivered to its collocated equipment. All power issues, including random inspections, acceptable buffer zones, and required annual attestations, are addressed in Verizon's tariff section 19.4.2.C.

A.

17 Q. IF POWER IS CHARGED ON A PER-AMP-USED BASIS, HOW 18 SHOULD THE CHARGE BE CALCULATED AND APPLIED? (ISSUE 19 6B)

Different companies may calculate and apply per-amp charges differently, but, for Verizon, the monthly recurring charge for DC Power should be calculated on a per-load-amp (as opposed to per-fused-amp) basis and should recover the following cost components: investment in installed power plant infrastructure, labor and material to extend cabling from power plant to Battery Distribution Fuse Bay (BDFB), fuses and

fuse panels on the BDFB, and an allocated utility cost. The per-amp charge should be applied for each load amp ordered by the ALEC.

Α.

4 Q. SHOULD THE ILEC INSTALL METERS TO MEASURE THE ACTUAL 5 AMPERAGE USED BY AN ALEC?

No. Placing meters in the central office to monitor usage on each cable feed is not feasible from a practical or cost standpoint. Metering would impose new costs on the ALEC because additional equipment would be introduced into the collocation configuration, along with additional manpower and administrative costs to read meters and bill accordingly. Verizon allows the ALEC to order only the power needed to operate its equipment when engineered using the manufacturers' specifications for industry-standard power drain. Today's digital equipment is designed to operate at a constant load, regardless of whether there is actually live traffic. In sum, metered power would raise costs and introduce inefficiency without yielding any advantages over Verizon's current practice.

A.

Q. WHEN SHOULD AN ILEC BE ALLOWED TO BEGIN BILLING AN ALEC FOR POWER? (ISSUE 6C)

Verizon begins billing the monthly recurring charges once the ALEC accepts the collocation space. Because part of Verizon's significant power investment is recovered in the per amp monthly charge, Verizon is entitled to begin recovery of that investment once the ALEC accepts the arrangement.

Q. SHOULD AN ALEC HAVE THE OPTION OF AN AC POWER FEED TO ITS COLLOCATION SPACE? (ISSUE 7)

No. Telecommunications equipment requires DC power, so the AC power from the electric utility must be converted into DC power to run the equipment. The ALEC may request additional AC power outlets to its collocation arrangement in order to operate various testing equipment or accommodate similar activities, but the ALEC should not be permitted to request AC power feeds with an intent to convert AC power to DC power within its collocation space. Conversion of AC power to DC power for telecommunications equipment is a core infrastructure function within the central office. Attempts to bypass this function by converting power within the collocation cage would require not only conversion equipment, but also batteries and generators, such that the ALEC has a backup power supply. In addition, special construction would typically be required to isolate the ALEC's power plant from surrounding equipment and protect against the risk of fire from the battery plant. Multiple power plants within the building multiply the risk from fire and/or hazardous materials. In order to maintain the safety and integrity of the network, the conversion of AC power to DC power should be performed at the central office power plant and then distributed to various points in the central office with associated power cabling and BDFBs.

24

25

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

A.

Q. WHAT ARE THE RESPONSIBILITIES OF THE ILEC, IF ANY, WHEN

1		AN ALEC REQUESTS COLLOCATION SPACE AT A REMOTE
2		TERMINAL WHERE SPACE IS NOT AVAILABLE OR SPACE IS
3		NEARING EXHAUSTION? (ISSUE 8)
4	A.	The procedures for obtaining collocation space at a remote terminal
5		should mirror those for a central office. The ILEC is not required to
6		construct additional space to satisfy a collocation request. If there is no
7		available space within the remote terminal, the ALEC should explore an
8		adjacent solution, such as placing its own remote terminal adjacent to
9		Verizon's terminal and establishing a network interface.
10		
11	Q.	PLEASE SUMMARIZE YOUR TESTIMONY.
12	A.	Verizon's intrastate collocation tariff has been in effect almost 3 years.
13		That tariff builds on longstanding experience with both interstate and
14		intrastate collocation arrangements. Verizon has approximately 250
15		collocation arrangements in service in Florida. Verizon's existing tariff
16		terms should remain in place because they are working well and they
17		adequately address the issues raised for resolution in this proceeding.
18		
19	Q.	DOES THIS CONCLUDE YOUR TESTIMONY?
20	A.	Yes.
21		
22		
23		
24		
25		

Docket Nos. 981834-TP & 990321-TP
John Ries Exhibit No. JR-1
FPSC Exhibit No.
Page 1 of 65

VERIZON FLORIDA INC.

FACILITIES FOR INTRASTATE ACCESS TARIFF

19. COLLOCATION SERVICE

EFFECTIVE: July 12, 2001 ISSUED: June 27, 2001

19. COLLOCATION SERVICE

CONTENTS

		Page No.	
19.1	General	1	
19.2	Description of Types of Collocation	1	
19.3	Ordering Conditions	4	
19.4	Installation and Operation	8	
19.5	Space Requirements	16	
19.6	Liability and Indemnification	20	(T)
19.7	<u>Insurance</u>	21	
19.8	Confidentiality	23	
19.9	<u>Casualty</u>	24	
19.10	Termination of Service	25	
19.11	Miscellaneous	26	
19.12	Virtual Collocation	26.1	(N)
19.13	Microwave Collocation	26.9	(N)
19.14	Rate Regulations	26.12	(T)
19.15	Description and Application of Rate Elements	26.12	(T)
19.16	Rates and Charges	35	(Τ)

Sixth Revised Page 1 Canceling Fifth Revised Page 1

19. COLLOCATION SERVICE

19.1 General

- Verizon (hereafter referred to as the Company) shall provide collocation services in accordance with, and subject to, the terms and conditions of this Tariff and any additional applicable regulations in other Company tariffs. The Company shall provide collocation services under this Tariff only to those parties which have an effective interconnection agreement with the Company for this state under Sections 251 and 252 of the Telecommunications Act of 1996, or have adopted such an agreement pursuant to Section 252(i) thereof. Requesting carriers may also seek to negotiate rates, terms, and conditions that are in addition to, or different from, the rates, terms, and conditions in this Tariff to the extent permitted by applicable law. As required by applicable law, the Company shall also offer rates, terms, and conditions for collocation services that are not expressly addressed in this Tariff or other Company tariffs on Bona Fide Request ("BFR") basis, and in doing so, shall comply with all applicable federal or state requirements. By agreeing to the rates, terms, and conditions of this Tariff or the collocation of any equipment hereunder: (1) the Company does not waive, and expressly reserves, its rights to continue to challenge the legality of the FCC Collocation Order (Docket No. 98-147) and to take further action regarding this matter as future circumstances warrant; (2) the Company does not intend to, and therefore does not, establish any precedent. waiver, course of dealing or in any way evidence the Company's position or intent with regard to future collocation requests; and (3) the Company specifically reserves the right to incorporate herein the decision by the United States Court of Appeals for the District of Columbia Circuit on March 17, 2000, and any other subsequent court (C) decisions affecting rules adopted by the FCC to implement collocation under Section 251 of the Telecommunications Act of 1996 (See, GTE Service Corporation, et. al. v. Federal Communications Commission (C) and United States of America, No. 99-1176, consolidated with No. 99-1201, 2000 U.S. App. LEXIS 4111 (D.C. Cir. 2000)).
- 19.1.2 Collocation provides for access to the Company's premises for the purpose of interconnection and/or access to unbundled network elements, including, its central offices and serving wire centers and all other buildings or similar structures owned, leased, or otherwise controlled by the Company that house the Company's network facilities, including but not limited to the Company's network facilities on public rights-of-way or in controlled environmental vaults (CEVs).
- 19.1.3 Collocation shall be accomplished through caged, cageless, virtual, or microwave collocation, except in those instances where not practical for technical reasons or due to space limitations. In such cases, the Company shall provide Adjacent Collocation or other methods of collocation, subject to space availability and technical feasibility.
- 19.1.4 The provision of collocation by the Company, as set forth in this Tariff, does not constitute a joint undertaking with the Competitive Local Exchange Carrier (CLEC) for the furnishing of the services. In addition, the regulations, terms and conditions of this Tariff do not apply to any CLEC offering of services to its subscribers.

19.2 Description of Types of Collocation

19.2.1 Single Caged

A single caged arrangement is a form of caged collocation, which allows a single CLEC to lease caged floor space to house its equipment within a Company premises. Additional details on single caged collocation are set forth in the Company's Collocation Services Packet, described in Section 19.3.1.

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

EFFECTIVE: October 13, 2001 ISSUED: September 28, 2001

Third Revised Page 2 Canceling Second Revised Page 2

19. COLLOCATION SERVICE

19.2 <u>Description of Types of Collocation</u> (Continued)

19.2.2 **Shared Caged**

A shared caged arrangement is a newly constructed caged collocation arrangement that is jointly applied for and occupied by two or more CLECs within a Company premises.

(D)

(C)

(C) (M)

(M)

(M) (C)

(D) The CLECs are solely responsible for determining whether to share a shared cage collocation arrangement and if so, upon what terms and conditions. The CLECs must each be interconnected to the Company for the exchange of (N) traffic with the Company and/or to access unbundled network elements. The Company will not issue separate billing for any of the rate elements associated with the shared caged collocation arrangement between the CLECs, but the Company will provide the CLECs with information on the proportionate share of the non-recurring charges for each CLEC in the shared arrangement. One of the CLECs in the shared cage collocation arrangements must be designated as the 'Billed CLEC" who will be billed and responsible for payment of all collocation applicable services ordered by the CLECs. Each CLEC will be responsible for ordering their own unbundled network elements from the Company. The Company will separately bill the CLECs for unbundled network elements ordered. (N)

Each CLEC in a shared collocation arrangement is the Company's customer and has all the rights and obligations (C) (M) applicable hereunder to CLECs purchasing collocation-related services, including, without limitation, the obligation to pay all applicable charges, whether or not the Billed CLEC is reimbursed for all or any portion of such charges by the other CLECs. All terms and conditions for caged collocation as described in this Tariff will apply to shared caged collocation requirements. Additional details on shared caged collocation are set forth in the Company's Collocation Service Packet, described in Section 19.3.1.

Subleased Caged 19.2.3

Vacant space available in a CLEC's existing caged collocation arrangement may be made available to a third party for the purpose of interconnection and/or for access to unbundled network elements in the Company's premises via the subleasing collocation arrangement. Details of subleasing collocation arrangements are set forth in the Company's Collocation Support Packet. The CLEC subleases floor space to the third party pursuant to terms and conditions agreed to by the CLEC and third party involved. The CLEC and third party must each be interconnected to the Company for the exchange of traffic with the Company and/or to access unbundled network elements.

The CLEC is solely responsible for determining whether to sublease a caged collocation arrangement and if so, (N) upon what terms and conditions. The Company will not issue separate billing for any of the rate elements associated with the subleased caged collocation arrangements between the CLEC and the third party. The CLEC will be responsible for payment of all collocation applicable services ordered by the CLEC and the third (N) party/parties.

EFFECTIVE: July 8, 2000

ISSUED: June 23, 2000

Certain material appearing on this page formerly appeared on second revised Page 1.

(N)

Second Revised Page 3
Canceling First Revised Page 3

19. COLLOCATION SERVICE

19.2 <u>Description of Types of Collocation</u> (Continued)

19.2.3 <u>Subleased Caged</u> (Continued)

Each CLEC and third party will be responsible for ordering their own unbundled network elements from the Company. The Company will separately bill the CLEC and third party/parties for unbundled network elements ordered. The CLEC and third party/parties are the Company's customers and have all the rights and obligations applicable hereunder to CLEC customers purchasing collocation-related services, including, without limitation, the obligation to pay all applicable charges, whether or not the CLEC is reimbursed for all or any portion of such charges by the third party/parties. All terms and conditions for caged collocation as described in this Tariff will apply to subleased caged collocation requirements. Additional details on subleased caged collocation are set forth in the Company's Collocation Services Packet, described in Section 19.3.1.

19.2.4 Cageless

Cageless collocation is a form of collocation in which CLEC's can place their equipment in Company premises space. A cageless collocation arrangement allows a CLEC, using Company approved vendors, to install equipment in single bay increments. The equipment location will be designated by the Company and will vary based on individual premises configurations. CLEC equipment will not share the same equipment bays with Company equipment. Additional details on cageless collocation are set forth in the Company's Collocation Services Packet, described in Section 19.3.1.

19.2.5 Adjacent

An adjacent collocation arrangement permits a CLEC to construct or procure a structure on Company property for collocation for the purposes of provisioning expanded interconnection and/or access to unbundled network elements in accordance with the terms and conditions of this Tariff. Adjacent collocation is only an option when the following conditions are met:

- Space is legitimately exhausted in the Company's premises for caged and cageless collocation; and
- It is technically feasible to construct or procure a hut, controlled environmental vault, or similar structure on Company property that adheres to local building code, zoning requirements, and Company building standards.

The CLEC is responsible for complying with all zoning requirements, any federal, state or local regulations ordinances and laws and obtaining all associated permits. The Company may, where required, participate in the zoning approval and permit acquisitions. The CLEC may not take any action in establishing an adjacent structure that will force the Company to violate any zoning requirements or any federal, state or local regulations ordinances, or laws.

Any construction by the CLEC on Company property must comply with Company technical specifications as they relate to environmental safety and grounding requirements set forth in the Company's Collocation Services Packet. The Company will make available power and physical collocation services to the CLEC in the same non-discriminatory manner as it provides to itself for the Company's Own Remote Equipment Buildings (REB).

Additional details on adjacent collocation are set forth in the Company's Collocation Services Packet, described in Section 19.3.1.

19.2.6 Virtual (T)

Under virtual collocation, the Telephone Company installs and maintains CLEC provided equipment which is dedicated to (C) the exclusive use of the CLEC in a collocation arrangement. Additional details on virtual collocation are set forth in Section 19.12. (C)

19.2.7 <u>Microwave</u> (N)

Physical collocation of microwave transmission facilities will be permitted on a first-come, first-served basis except where (C) such collocation is not practical for technical reasons or because of space limitations. Microwave collocation provides for the interconnection of CLEC or Company provided facilities equipment and support structures located in, on or above the exterior walls and roof of Company premises. Additional details on Microwave Collocation are set forth in Section 19.13.

(D)

EFFECTIVE: July 12, 2001

ISSUED: June 27, 2001

Second Revised Page 4 Canceling First Revised Page 4

 \mathbf{m}

19. COLLOCATION SERVICE

19.3 **Ordering Conditions**

19.3.1 **Application**

(A) Point of Contact/Collocation Services Packet

The Company will establish points of contact for the CLEC to contact to place a request for collocation. The point of contact will provide the CLEC with the Collocation Services Packet, which shall contain general information and requirements, including a list of engineering and technical specifications, fire, safety, security policies and procedures, and an application form.

Application Form/Fee (B)

CLECs requesting collocation at a premises will be required to complete the application form and submit the non-refundable engineering fee set forth in Section 19.16 for each premises at which collocation is (T) requested. The application form will require the CLEC to provide all engineering, floor space, power, environmental and other requirements necessary for the function of the service. The CLEC will also provide the Company with specifications for any non-standard or special requirements at the time of application. The Company reserves the right to assess the CLEC any additional charges not included in Section 19.16 on an Individual Case Basis (ICB) associated with complying with the application request.

(C) Notification of Acceptance/Rejection

The Company will notify the CLEC in writing within fifteen (15) calendar days following receipt of the completed application if the CLEC's requirements cannot be accommodated as specified. Should the CLEC submit ten (10) or more applications within a ten (10) day period, the response interval will be increased as follows: Interval for Applications 10-19: Within 25 calendar days from receipt of the first application; Interval for Applications 20-29: Within 35 calendar days from receipt of the first application; Interval for 30+ applications: 10 additional calendar days (after 35 calendar days from receipt of the first application) for each 10 additional applications.

(D) Changes

The first application form submitted by the CLEC shall be designated the original application. Original applications for collocation arrangements that have not been inspected and approved by the CLEC are subject to requests for minor or major changes to the services requested in the application. Changes will not be initiated until a completed application has been submitted along with the appropriate Engineering Fee, if applicable.

Major changes are requests that add telecommunications equipment that requires additional AC or DC power systems; heating, ventilation and air conditioning (HVAC) system modifications; or change the size of the cage. At the election of the CLEC, major changes may be handled in one of the following two methods to the extent technically feasible.

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

EFFECTIVE: July 12, 2001 ISSUED: June 27, 2001

(M)

19. COLLOCATION SERVICE

19.3. Ordering Conditions (Continued)

19.3.1 Application (Continued)

(D) Changes (Continued)

(1) Method 1: Additional Application

The CLEC may elect to have a major change to its original collocation application treated by the Company as an additional (new) application. An additional application is subject to the same provisioning process and conditions as an original application. On receipt of a complete additional application and Engineering Fee, the Company will notify the CLEC in writing within fifteen (15) calendar days following receipt of the completed additional application if the CLEC's additional requirements cannot be accommodated as specified. Filing an additional application does not change the Company's obligation to process and fulfill the original application nor does it change the time intervals applicable to the processing and fulfillment of the original application. All of the provisions herein applicable to an original application similarly apply to an additional application.

(2) Method 2: Supplemental Application

The CLEC may elect to have a major change to its original collocation application treated by the Company as a supplemental application. A supplemental application may affect the Company's obligation to process and fulfill the original application. On receipt of a supplemental application and Engineering Fee, the Company will notify the CLEC in writing within fifteen (15) calendar days following receipt of the completed supplemental application if the CLEC's requirements cannot be accommodated as specified. Upon notification that the Company can accommodate the requirements of the supplemental application, the CLEC may elect to proceed with the supplemental application. The Company's obligations under the original application will be merged with the obligations of the supplemental application and the combined project time line will be based on the date the supplemental application was received. All of the provisions herein applicable to an original application similarly apply to a supplemental application.

Minor changes are those requests that do not require additional AC or DC power systems, HVAC system upgrades, or changes in cage space. The CLEC will be required to submit a revised application, but the deliverable dates for the project will not change.

EFFECTIVE: July 8, 2000

ISSUED: June 23, 2000

(M)

19. COLLOCATION SERVICE

19.3 **Ordering Conditions (Continued)**

19.3.2 Space Availability

The Company will notify the CLEC within fifteen (15) calendar days following receipt of the completed application form and non-refundable engineering fee, if space is available at the selected premises. If no space is available, the Company will notify the CLEC in writing. If only partial collocation space is available to accommodate the CLEC request, the Company will specify the amount of available collocation space to the CLEC in writing. Space availability and reservation shall be determined in accordance with Section 19.5 following.

19.3.3 **Price Quote**

The Company shall provide the CLEC with a detailed price quote for collocation services required to accommodate the CLEC's request within fifteen (15) calendar days of the application date, provided that the application form is complete and accurate and no Individual Case Basis (ICB) rates are required in the quote. If the Company determines that the application form is incomplete or defective, the Company will inform the CLEC as soon as possible and will identify which sections of the application form are incomplete or defective. The quote will be honored for ninety (90) calendar days from the date of issuance. However, the Company reserves the right to change the price quote at any time prior to acceptance by the CLEC. If the quote is not accepted by the CLEC within such ninety (90) day period, the CLEC will be required to submit a new application form and engineering fee and a new quote will be provided based on the new application form.

19.3.4 Access Service Request (ASR)

Upon notification of available space, the CLEC will be required to send a completed Access Service Request (ASR) form to the Company's collocation point of contact. A copy of an ASR form is included in the Collocation Services Packet.

19.3.5 **Augmentation**

All requests for an addition or change to an existing collocation arrangement that has been inspected and turned over to the CLEC is considered an augmentation. An augmentation request will require the submission of a complete application form and a non-refundable Engineering or Minor Augment fee. A Minor Augment fee may not be required under certain the circumstances outlined below. The definition of a major or minor augment is as follows:

Major Augments are those requests that require additional AC or DC power, HVAC system upgrade(s), or A) floor space. A complete application and Engineering Fee will be required when submitting a request that (T) requires a major augment. (T)

EFFECTIVE: July 12, 2001

ISSUED: June 27, 2001

Fourth Revised Page 7
Canceling Third Revised Page 7

EFFECTIVE: October 13, 2001

ISSUED: September 28, 2001

19. COLLOCATION SERVICE

19.3 Ordering Conditions (Continued)

19.3.5 Augmentation (Continued)

(B) Minor Augments of collocation arrangements will require the submission of a complete application form and the Augment Fee. Minor augments are those requests that do not require more AC or DC power, HVAC system upgrades, or additional floor space. The requirements for a minor augment request can not exceed the capacity of the existing electrical/power or HVAC system. Requests for additional DSO, DS1, and DS3 cross connects for access to unbundled network elements are included as minor augments.

(C)

Minor augments that require an augment fee are those requests that require the Company to perform a service or function on behalf of the CLEC including but not limited to: installations of Virtual equipment cards or software upgrades, removal of Virtual equipment, requests to pull cable from exterior microwave facilities and requests to terminate DSO, DS1, and DS3 cables.

(C)

Minor augments that do not require a fee are those augments performed solely by the CLEC, that do not require the Company to provide a service or function on behalf of the CLEC, including but not limited to, requests to install additional equipment in the CLEC's cage which do not exceed the power and HVAC engineering specifications for the existing arrangement. Prior to the installation of the additional equipment the CLEC agrees to provide the Company with an application form with an updated equipment listing that includes the equipment to be installed in the CLEC's collocation arrangement. Once the updated equipment list is submitted to the Company, the CLEC may proceed with the augment. The CLEC agrees that changes in equipment provided by the CLEC under this provision will not exceed the engineering specifications for power and HVAC as requested on the original application. All augments will be subject to Company inspection, in accordance with the terms of this Tariff for the purpose of ensuring compliance with Company safety standards.

The Company will notify the CLEC in writing within fifteen (15) calendar days following receipt of the completed augment application if the CLEC's requirements cannot be accommodated as specified. Should the CLEC submit ten (10) or more augment applications within a ten (10) day period, the response interval will be increased as follows: Interval for Applications 10-19: Within 25 calendar days from receipt of the first application; Interval for Applications 20-29: Within 35 calendar days from receipt of the first application, Interval for 30+ applications: 10 additional calendar days (after 35 calendar days from receipt of the first application) for each 10 additional applications.

The Company shall provide the CLEC with a detailed price quote for collocation services required to accommodate the CLEC's augment request within fifteen (15) calendar days of the application date, provided that the augment application form is complete and accurate and no Individual Case Basis (ICB) rates are required in the quote. If the Company determines that the augment application form is incomplete or defective, the Company will inform the CLEC as soon as possible and will identify which sections of the augment application form are incomplete or defective. The augment quote will be honored for ninety (90) calendar days from the date of issuance. However, the Company reserves the right to change the augment price quote at any time prior to acceptance by the CLEC. If the augment quote is not accepted by the CLEC within such ninety (90) day period, the CLEC will be required to submit a new augment application form and engineering fee and a new quote will be provided based on the new augment application form.

19.3.6 Expansion

The Company will not be required to construct additional space to provide for caged, cageless and/or adjacent collocation when available space has been exhausted. Where the CLEC seeks to expand its existing collocation space, the Company shall make contiguous space available to it to the extent possible; provided, however, the Company does not guarantee contiguous space to the CLEC to expand its existing collocation space. CLEC requests for expansion of existing space within a specific premises will require the submission of an application form and the appropriate major augment fee.

19. COLLOCATION SERVICE

19.3.7 Relocation

(M)

CLEC requests for relocation of the termination equipment from one location to a different location within the same premises will be handled on an ICB basis. The CLEC will be responsible for all costs associated with the relocation of its equipment.

(C)(M)

 $(D)^*$

19.3.8 **Conversions**

Requests for converting Virtual Collocation arrangements to Caged or Cageless arrangements shall be submitted and (N) designated as an Augment Application described in Section 19.3.5. Requests for converting a Virtual arrangement to a Cageless arrangement which requires no physical changes to the arrangement will be assessed a Minor Augment fee. All other conversion requests for Virtual to Caged or Cageless will be assessed an Engineering/Major Augment Fee and other applicable charges. The Company will notify the CLEC within fifteen (15) calendar days following receipt of the completed Augment Application if the CLEC conversion request is accepted or denied. When converting a Virtual arrangement to a Caged or Cageless arrangement, the CLEC's equipment may need to be relocated. The CLEC will be responsible for all costs associated with the relocation of its equipment as described in Section 19.3.7.

(N)

Certain material appearing on this page formerly appeared on Original Page 4. (D)* - Removed duplicate section 19.3.7.

(N)

(N)

Fourth Revised Page 8 Canceling Third Revised Page 8

19. COLLOCATION SERVICE

19.4 Installation and Operation

19.4.1 Planning and Coordination

Upon receipt of the ASR and fifty percent (50%) of the applicable nonrecurring Charges (NRCs), set forth in Section 19.16, associated with the ordered collocation services, the Company will:

- (A) Schedule a meeting with the CLEC to determine engineering and network requirements.
- (B) Initiate the necessary modifications to the premises to accommodate the CLEC's request.
- (C) Work cooperatively with the CLEC to ensure that services are installed in accordance with the service requested.

The CLEC is responsible for coordinating with the Company to ensure that services are installed in accordance with the ASR. The CLEC shall meet with the Company, if requested by the Company, to review design and work plans for installation of CLEC designated equipment within the Company's premises. The CLEC is responsible to have all of its cables and other equipment ready for installation on the date scheduled. If the CLEC fails to notify the Company of a delay in the installation date, it will be subject to the appropriate labor charge(s) set forth in Section 19.16. The Company will permit the CLEC to schedule one escorted visit to the CLEC's collocation space during construction. The labor rates in Section 19.16 will be applied for the escorted visit.

19.4.2 Space Preparation

(A) Cage Construction

For caged collocation, the Company will construct the cage with a standard enclosure or the CLEC may subcontract this work to a Company approved contractor.

(B) Site Selection/Power

The Company shall designate the space within its premises where the CLEC shall collocate its equipment. The Company will assign collocation space to the CLEC in a just, reasonable, and nondiscriminatory (N) manner. The Company will allow the CLEC requesting caged or cageless collocation to submit space preferences on the Application Form prior to assigning caged and cageless collocation space to the CLEC. The Company will assign caged and cageless space in accordance with the following standards: (1) The CLEC's collocation costs cannot be materially increased by the assignment; (2) The CLEC's occupation and use of the Company's premises cannot be materially delayed by the assignment; (3) The assignment cannot impair the quality of service or impose other limitations on the service the CLEC wishes to offer; and (4) The assignment cannot reduce unreasonably the total space available for caged and cageless collocation, or preclude unreasonably, caged and cageless collocation within the Company's premises.

The Company may assign caged and cageless collocation space separate from space housing the Company's equipment, provided that each of the following conditions is met: (1) Either legitimate security concerns, or operational constraints unrelated to the Company's or any of its affiliates' or subsidiaries competitive concerns, warrant such separation; (2) Any caged and cageless collocation space assigned to an affiliate or subsidiary of the Company is separated from space housing the Company's equipment; (3) The separated space will be available in the same time frame as, or a shorter time frame than, non-separated space; (4) The cost of the separated space to the CLEC will not be materially higher than the cost of non-separated space; and (5) The separated space is comparable, from a technical and engineering standpoint, to non-separated space.

(N)

(M)

(M) Material moved to Page 8.1.

(N)

Third Revised Page 8.1 Canceling Second Revised Page 8.1

19. COLLOCATION SERVICE

19.4 <u>Installation and Operation (Continued)</u>

19.4.2 Space Preparation (Continued)

(B) Site Selection/Power (Continued)

The Company shall provide, at the rates set forth in Section 19.16, 48V DC power with generator and/or battery back-up, AC convenience outlet, heat, air conditioning and other environmental support to the CLEC equipment in the same standards and parameters required for Company equipment within that premises. The Company will be responsible for the installation of the AC convenience outlets, overhead lighting and equipment superstructure per the established rates.

(C) DC Power

The Company will provide DC power to the collocation arrangement as specified by the CLEC in its Collocation application. The CLEC will specify the load on each feed and the size of the fuse to be placed on each feed. CLECs must order a minimum of ten (10) load amps for each caged, cageless, and virtual collocation arrangement. CLECs may order additional DC Power (beyond the minimum) in one (1) amp increments. Charges for DC power will be applied based on the total number of load amps ordered on each feed.

For example, if a CLEC orders a total of 40 load amps of DC power and an A and B feed, the CLEC could order 20 load amps on the A feed and 20 load amps on the B feed. The Company will permit the CLEC to order a fuse size up to 2.5 times the load amps ordered provided that applicable law permits this practice. Thus, the CLEC could order that each feed be fused at 50 amps if the CLEC wants one feed to carry the entire load in the event the other feed fails. Accordingly, the CLEC will be charged on the basis of the total number of load amps ordered, i.e., 40 amps, and not based on the total number of amps available for the fuse size ordered.

The Company reserves the right to perform random inspections to verify the actual power load being drawn by a Collocation arrangement. At any time, without written notice, the Company may measure the DC power drawn at an arrangement by monitoring the Company's power distribution point. In those instances where the Company needs access to the collocation arrangement to make these measurements, the Company will schedule a joint meeting with the CLEC.

- (1) If the inspection reveals that the power being drawn does not exceed the total number of load amps ordered, no further action will apply.
- (2) If the inspection reveals that the power being drawn is within the applicable buffer zone, as defined in this subsection, that arrangement is subject to the following treatment:
 - (a) The Company will provide the CLEC with written notification, by certified US mail to the person designated by the CLEC to receive such notice, that more power is being drawn than was ordered. Within ten (10) business days of the date of receipt of notification, the CLEC must reduce the power being drawn to match its ordered load or revise its power requirement to accommodate the additional power being drawn. The Company will accept a certification signed by a representative of the CLEC that power consumption has been reduced to match the ordered load. Failure to reduce the power being drawn or submit a revised application within ten (10) business days will result in an increase in the amount of power being billed to the audited load amount.
 - (b) For a collocation arrangement that has 100 amps or less fused, the buffer zone for the first two violations during a consecutive twelve (12) month period will be 120% of load, as long as the second violation is not for the same collocation arrangement as the first. For any subsequent violations, or if the second violation is for the same collocation arrangement, and for any violation where the collocation arrangement has more than 100 amps fused, the buffer zone will be 110% of load.

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

EFFECTIVE: October 1, 2002 ISSUED: September 16, 2002

19. COLLOCATION SERVICE

19.4 <u>Installation and Operation (Continued)</u>

(N)

19.4.2 Space Preparation (Continued)

(C) DC Power (Continued)

- (3) If the first inspection reveals that the power being drawn is greater than the applicable buffer zone specified in 2.b. preceding, that arrangement is subject to the following treatment:
 - (a) The Company will notify the person designated by the CLEC to receive such notice via telephone or e-mail that the Company will take a second measurement no sooner than one (1) hour and no later than two (2) days after the initial inspection. The Company will not wait for the CLEC or require it to be present during the second inspection.
 - (b) The Company will assess a nonrecurring charge for the additional labor to perform this inspection. The nonrecurring charge applies for the first half hour (or fraction thereof) and for each additional half hour (or fraction thereof) per technician, per occurrence as shown in 19.16.51 following.
 - (c) The CLEC may perform its own inspection at the CLEC's cage. The CLEC is not required to wait for the Company or require it to be present during the CLEC test. Upon request of the CLEC, the Company will send a representative to accompany the CLEC to conduct a joint inspection at the CLEC cage at no charge to the CLEC. Nothing herein shall be construed to prohibit the CLEC from testing at its own cage. The CLEC will send the results of its own audit measurements to the Company if they are taken in response to a notice of violation under this section and if the CLEC's measurements differ from the Company's.
 - (d) If the second test also exceeds the applicable buffer zone, the Company will provide the CLEC with written notification, within ten (10) business days, by certified U.S. mail to the person designated by the CLEC to receive such notice that it has exceeded its ordered power. The notification will include: (1) initials or identifying number of the Company technician(s) who performed the inspection; (2) dates and times of the inspections; (3) the make, model and type of test equipment used; (4) the length of monitoring and the results of the specific audit; (5) the total load amps currently being billed; (6) how the test was done; and (7) any other relevant information or documents.
 - (e) The Company will maintain a file of results taken of any inspections for two (2) years and such file will be made available to the CLEC that was audited, upon request. The Company will treat as confidential information the identity of CLECs that it audits as well as the results of such audits, unless it receives prior written consent of the affected CLEC to disclose such information. The foregoing does not preclude the Company from making the notice described in paragraph f. following.
 - (f) If the CLEC disagrees with the results of the audit, it will first notify the Company. The Company and the CLEC will make a good faith effort to resolve the issue. If the parties do not resolve the issue, either party can invoke dispute resolution processes in the applicable interconnection agreement. The dispute resolution process can be initiated by either party after thirty (30) calendar days have elapsed. This period commences: (1) ten (10) business days from receipt of the notification, in the case of a violation within the buffer zone; or (2) after the CLEC has received notice of the second test, in the case of a violation over the buffer zone.
 - (g) With the notification required by subparagraph C.3.d., the Company will also notify the CLEC that it must submit a non-scheduled attestation of the power being drawn at each of its remaining collocation arrangements. The CLEC must submit this non-scheduled attestation within fifteen (15) business days of the date of this notification. Failure to submit this non-scheduled attestation will result in the application of additional labor charges for any subsequent DC power inspections the Company performs prior to receipt of the next scheduled attestation. Scheduled attestations are defined in this section.

(N)

19. COLLOCATION SERVICE

19.4 Installation and Operation (Continued)

(N)

19.4.2 Space Preparation (Continued)

- (C) DC Power (Continued)
 - (4) If the inspection reveals that the power being drawn is greater than the applicable buffer zone, then the penalty will be as follows.
 - (a) For the first such violation within the same consecutive twelve (12) month period, the CLEC will be billed the audited load amount for four (4) months. The CLEC will pay a separate and additional penalty to the American Red Cross, measured as the difference between the billing of the fused capacity and the billing at the audited load for four (4) months. The CLEC must send notice of its American Red Cross payment to the Company within ten (10) calendar days of making the payment.
 - (b) For the second such violation within the same consecutive twelve (12) month period, the CLEC will be billed the audited load amount for five (5) months. The CLEC will pay a separate and additional penalty to the American Red Cross, measured as the difference between the billing of the fused capacity and the billing at the audited load for five (5) months. The CLEC must send notice of its American Red Cross payment to the Company within ten (10) calendar days of making the payment.
 - (c) For the third such violation within the same consecutive twelve (12) month period, the CLEC will be billed the audited load amount for six (6) months. The CLEC will pay a separate and additional penalty to the American Red Cross, measured as the difference between the billing of the fused capacity and the billing at the audited load for six (6) months. The CLEC must send notice of its American Red Cross payment to the Company within ten (10) calendar days of making the payment.
 - (d) For more than three (3) violations within the same consecutive twelve (12) month period, the Company will bill at the fused amount for a minimum of six (6) months and continue to bill at the fused amount until an updated attestation or augment specifying revised power is received, and nothing will be paid to the American Red Cross.
 - (e) The Company will notify the CLEC that it is being billed under a penalty situation, designating the applicable number of months and also calculating the penalty owed to the American Red Cross, under the provisions set forth preceding.
 - (5) At the conclusion of any dispute resolution proceeding, the above penalties (including the revised billing) will be self-executing.
 - (6) If the CLEC has requested a power augment under which the audited amount would be within the augmented load, plus the applicable buffer zone, and the augment is late due to the fault of the Company, the penalty will not be imposed and the parties will not count this instance for purposes of determining what type of penalty to impose under 4e. preceding.

(N)

Second Revised Page 8.2 Canceling First Revised Page 8.2

19. COLLOCATION SERVICE

19.4 <u>Installation and Operation (Continued)</u>

19.4.2 <u>Space Preparation (Continued)</u>

(C) <u>DC Power</u> (Continued)

(D)

(T)

(Ť)

(D)

Annually, each CLEC must submit a written statement signed by a responsible officer of the Company, which attests that it is not exceeding the total load of power as ordered in its Collocation applications. This attestation, which must be received by the Company no later than the last day of June, shall individually list all of the CLEC's completed Collocation arrangements provided by the Company in the state. If the CLEC fails to submit this written statement by the last day in June, the Company will notify the CLEC in writing that it has thirty (30) calendar days to submit its power attestation. Failure to submit the required statement within the 30 calendar day notice period will result in the billing of DC power at each Collocation arrangement to be increased to the total number of amps fused.

Whenever the Company is required to perform work on an Collocation arrangement as a result of a CLEC's order for a reduction in power requirements (e.g., change in fuse size), the Company will assess a nonrecurring charge for the additional labor. The nonrecurring charge applies for the first half hour (or fraction thereof) and for each additional half hour (or fraction thereof) per technician, per occurrence as shown in Section 19.16.51 following.

If the CLEC orders a change in the power configuration requiring new -48 volt DC power feeds to the Collocation arrangement, the Company will require an Engineering/Major Augment Fee with an application as set forth in Section 19.16.1 following, subject to the terms and conditions described in Section 19.3.5. In addition, if a CLEC's order for a reduction in DC power triggers the deployment of power cabling to a different power distribution point, the Engineering/Major Augment Fee as set forth in Section 19.16.1 following applies. The Company will work cooperatively with the CLEC to configure the new power distribution cables and disconnect the old ones.

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

EFFECTIVE: February 15, 2002 ISSUED: January 31, 2002

Fourth Revised Page 9 Canceling Third Revised Page 9

19. COLLOCATION SERVICE

19.4 <u>Installation and Operation (Continued)</u>

19.4.2 Space Preparation (Continued)

(D) <u>Timing</u>

The Company shall use its best efforts to minimize the additional time required to condition collocation space, and will inform the CLEC of the time estimates as soon as possible. For new caged and cageless arrangements, the Company shall complete delivery of the floor space to the CLEC within ninety (90) calendar days of receipt of fifty percent (50%) of the NRCs, assuming that the material shipment and construction intervals for the improvements required to accommodate the request (e.g., HVAC, system/power plant upgrade/ cables) are met. If the Company is unable to accommodate the CLEC caged or cageless request within the applicable number of days and the CLEC does not agree to an extension, the Company shall submit a Motion for Extension of Time to the Commission and CLEC within 45 calendar days of receipt of fifty percent (50%) of the NRCs. The Motion shall explain, in detail, the reasons necessitating the extension.

For augments to existing Caged and Cageless arrangements, the Company shall complete delivery of the floor space to the CLEC within forty-five (45) calendar days of receipt of fifty percent (50%) of the NRCs, assuming that the material shipment and construction intervals for the improvements required to accommodate the request (e.g., HVAC, system/power plant upgrade/ cables) are met. If the Company is unable to accommodate the CLEC's Caged or Cageless augment request within the applicable number of days and the CLEC does not agree to an extension, the Company shall submit a Motion for Extension of Time to the Commission and CLEC within 30 calendar days of receipt of fifty percent (50%) of the NRCs. The Motion shall explain, in detail, the reasons necessitating the extension.

Space delivery within such time frames shall also be subject to the permitting process of the local municipality. Prior to the CLEC beginning the installation of equipment in a cage, bay or cabinet, the CLEC and the Company must conduct a walk through of the designated collocation space. Upon acceptance of the arrangement by the CLEC, billing will be initiated, access cards will be issued and the CLEC may begin installation of equipment.

(E) Forecasting and Use of Data

The Company will request Cages and Cageless forecasts from the CLEC on an annual basis, with each forecast covering a two-year period. The CLEC will be required to update the near-term (6-month) forecasted application dates. Information requested will include central office, month applications are expected to be sent, requested in-service month, preference for Caged or Cageless collocation, and square footage required.

Unforecasted demand will be given a lesser priority than forecasted demand. The Company will make every attempt to meet standard intervals for unforecasted requests. However, if unanticipated requests push demand beyond the Company's capacity limits, the Company will negotiate longer intervals as required (and within reason). In general, if forecasts are received less than two (2) months prior to the application date, delivery of the floor space to the CLEC may be postponed by no more than sixty (60) calendar days.

19.4.3 Equipment and Facilities

(A) Purchase of Equipment

The CLEC will be responsible for supply, purchase, delivery, installation and maintenance of its equipment and equipment bay(s) in the collocation area. If the CLEC chooses, the Company will assist the CLEC in the purchase of equipment by establishing a contact point with Verizon Logistics, Inc. The Company is not responsible for the design, engineering, or performance of CLEC equipment and provided facilities for collocation. Upon installation of all transmission and power cables for collocation services, the CLEC relinquishes all rights, title and ownership of transmission (excluding fiber entrance facility cable) and power cables to the Company.

(M)

(M) Material moved to Page 10.

(N)

Third Revised Page 10 **Canceling Second Revised Page 10**

19. COLLOCATION SERVICE

19.4 Installation and Operation (Continued)

19.4.3 **Equipment and Facilities (Continued)**

(B) Permissible Equipment

(M)

The Company shall permit the collocation and use of any equipment necessary for interconnection or (N) access to unbundled network elements in accordance with the following standards: (1) Equipment is necessary for interconnection if an inability to deploy that equipment would, as a practical, economic, or operational matter, preclude the CLEC from obtaining interconnection with the Company at a level equal in quality to that which the Company obtains within its own network or the Company provides to any of its affiliates, subsidiaries, or other parties; and (2) Equipment is necessary for access to an unbundled network element if an inability to deploy that equipment would, as a practical, economic, or operational matter, preclude the CLEC from obtaining nondiscriminatory access to that unbundled network element, including any of its features, functions, or capabilities.

Multi-functional equipment shall be deemed necessary for interconnection or access to an unbundled network element if and only if the primary purpose and function of the equipment, as the CLEC seeks to deploy it, meets either or both of the standards set forth in the preceding paragraph. For a piece of equipment to be utilized primarily to obtain equal in quality interconnection or nondiscriminatory access to one or more unbundled network elements, there also must be a logical nexus between the additional functions the equipment would perform and the telecommunication services the CLEC seeks to provide to its customer by means of the interconnection or unbundled network element. The collocation of those functions of the equipment that, as stand-alone functions, do not meet either of the standards set forth in the preceding paragraph must not cause the equipment to significantly increase the burden on the Company's property.

Whenever the Company objects to collocation of equipment by a requesting CLEC for purposes within the scope of Section 251(c)(6) of the Act, the Company shall prove to the state commission that the equipment is not necessary for interconnection or access to unbundled network elements under the standards set forth above.

(N)

The CLEC may place in its collocation space ancillary equipment such as cross connect frames, and metal (M) storage cabinets. However, metal storage cabinets must meet Company premise environmental standards.

(M)

(C) **Specifications**

The CLEC facilities shall not physically, electronically, or inductively interfere with or impair the service of the Company's or any other CLEC facilities, create hazards or cause physical harm to any individual or the public. All CLEC equipment used for caged and cageless collocation must be tested to, and must meet: (1) the NEBS Level 1 family of safety requirements as defined in the Company's NEBS requirements RNSA-NEB-95-0003, Revision 10 or higher; and (2) the specific risk/safety/hazard criteria specified in Addendum E of the Collocation Services Packet. Any CLEC equipment that does not conform to these requirements may not be installed on Company property.

The Company reserves the right to remove and/or refuse use of CLEC facilities and equipment from its list of approved products if such products, facilities and equipment are determined to be no longer compliant with NEBS Level 1 safety standards or Electromagnetic Compatibility and Electrical Safety Generic Criteria for Network Telecommunication Equipment (GR-1089-CORE). The Company also reserves the right to remove and/or refuse use of CLEC facilities or equipment which does not meet or comply with the NEBS equipment safety requirements specified in Addendum E of the Collocation Services Packet.

(M) Material moved from Page 9.

(N)

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

EFFECTIVE: October 13, 2001 ISSUED: September 28, 2001

Second Revised Page 11 Canceling First Revised Page 11

EFFECTIVE: July 12, 2001 ISSUED: June 27, 2001

19. COLLOCATION SERVICE

19.4 <u>Installation and Operation (Continued)</u>

19.4.3 Equipment and Facilities (Continued)

(D) Cable

The CLEC is required to provide proper cabling, based on circuit type (VF, DS0, xDSL, DS1, DS3, etc.) to ensure adequate shielding. The Company cable standards (which are set forth in the Collocation Services Packet) are required to reduce the possibility of interference. The CLEC is responsible for providing fire retardant riser cable that meets Company standards. The Company is responsible for placing the CLEC's fire retardant riser cable from the cable vault to the collocation space. The Company is responsible for installing CLEC provided fiber optic cable in the cable space or conduit from the first manhole to the premises. This may be shared conduit with dedicated innerduct.

If the CLEC provides its own fiber optic facility, the CLEC shall be responsible for bringing its fiber optic cable to the premises manhole. The CLEC must leave sufficient cable length for the Company to be able to fully extend such cable through to the CLEC's collocation space.

(E) Manhole/Splicing Restrictions

The Company reserves the right to prohibit all equipment and facilities, other than fiber optic cable, from its entrance manholes. The CLEC will not be permitted to splice fiber optic cable in Manhole #1 (first Company manhole outside of the wire center). Where the CLEC is providing underground fiber optic cable in Manhole #1, it must be of sufficient length as specified by the Company to be pulled through the premises conduit to the CLEC collocation arrangement. The Company is responsible for installing a cable splice, if necessary, where CLEC provided fiber optic cable meets Company standards within the premises cable vault or designated splicing chamber. The Company will provide space and racking for the placement of an approved secured fire retardant splice enclosure.

(F) Access Points and Restrictions

The interconnection point for caged and cageless collocation is the point where collocation cable facilities (T) connect to Company termination equipment. The demarcation point for the CLEC is its terminal equipment or interconnect/cross connect panel within its cage, bay/frame or cabinet. The CLEC must tag all entrance facilities to indicate ownership. The CLEC will not be allowed access to Company DSX line-ups, MDF or any other Company facility termination points. The DSX, and MDF, and fiber distribution panel are to be considered Company demarcation points only. Only Company employees, agents or contractors will be allowed access to the MDF, DSX, or fiber distribution panel to terminate facilities, test connectivity, run (T) jumpers and/or hot patch in-service circuits.

Fourth Revised Page 12 Canceling Third Revised Page 12

19. COLLOCATION SERVICE

19.4 <u>Installation and Operation</u> (Continued)

19.4.3 Equipment and Facilities (Continued)

(G) Staging Area

For caged and cageless collocation arrangements, the CLEC shall have the right to use the designated staging area, a portion of the premises and loading areas, if available, on a temporary basis during its equipment installation work in the collocation space. The CLEC is responsible for protecting the Company's equipment and premises walls and flooring within the staging area and along the staging route. The CLEC will meet all Company fire, safety, security and environmental requirements. The temporary staging area will be vacated and delivered to the Company in an acceptable condition upon completion of the installation work. The CLEC may also utilize a staging trailer, which can be located on the exterior premises. The Company may assess the CLEC a market value lease rate for the area occupied by the trailer.

(H) Testing

Upon installation of the CLEC equipment, with prior notice, the Company will schedule time to work with the CLEC during the turn-up phase of the equipment to ensure proper functionality between CLEC equipment and the connections to Company equipment. The time period for this to occur will correspond to the Company's maintenance window installation requirements. The CLEC is solely responsible to provide its own monitor and test points, if required, for connection directly to their terminal equipment.

(I) Interconnection Between Collocated Spaces

(T)

(1) Dedicated Transit Service (DTS), which allows for interconnection between CLECs, provides a dedicated electrical or optical path between collocation arrangements (caged, cageless, and virtual) of the same or of two different CLECs within the same Company premises, using Company provided distribution facilities. DTS is available for DS0, DS1, DS3, and dark fiber cross connects. In addition, the Company will also provide other technically feasible cross-connection arrangements, including lit fiber, on an Individual Case Basis (ICB) as requested by a CLEC. The Company will offer DTS to requesting CLECs as long as such access is technically feasible.

DTS is only available when both collocation arrangements (either caged, cageless, and/or virtual) being interconnected are within the same Company premises, provided that the collocated equipment is used for interconnection with the Company and/or for access to the Company's unbundled network elements. The Company shall provide such DTS connections from the CLEC's collocation arrangement to another collocation arrangement of the same CLEC within the same Telephone Company premises, or to a collocation arrangement of another CLEC in the same Telephone Company premises. DTS is provided at the same transmission level from CLEC to CLEC.

The DTS arrangement requires the requesting CLEC to provide cable assignment information for itself as well as for the other CLEC. The Company will not make cable assignments for DTS. The requesting CLEC is responsible for all DTS ordering, bill payment, disconnect orders and maintenance transactions and is the customer of record. The requesting CLEC must also provide a letter of agency from the CLEC it is connecting to that authorizes the DTS connection and facility assignment. DTS is provided on a negotiated interval with the requesting CLEC. DTS service order and service connection rates are specified in Section 19.16.

(N)

(M)

(M) Material moved to Page 12.1.

(N)

JOHN P. BLANCHARD, PRESIDENT

TAMPA, FLORIDA

EFFECTIVE: October 13, 2001
ISSUED: September 28, 2001

19.4 **Installation and Operation (Continued)**

(N)

19.4.3 **Equipment and Facilities (Continued)**

(N)

(N)

(NOTE: Collocator to Collocator Interconnect arrangements will only be provided under the terms and conditions of this section to CLECs with existing arrangements. Additions, rearrangements, and moves are not permitted.)

(N)

The Company shall provide, upon the CLEC's request, a Collocator to Collocator Interconnect (M) arrangement between its equipment and the equipment of other collocated CLECs. When initiating a Collocator to Collocator Interconnect request, the CLEC must submit an Application Form, ASR, and a Minor Augment Fee. The Company will be responsible for engineering and installing the overhead superstructure for the Collocator to Collocator Interconnect arrangement, if required, and determining the appropriate cable route.

If the Collocator to Collocator Interconnect arrangement is between CLECs who occupy contiguous collocation cages or bays, an Application Form is required. The Minor Augment Fee will not be required if engineering and installation of overhead superstructure is not necessary to complete the arrangement.

The CLEC has the option of providing all cables and connectors for the arrangement and the option of running the cables for the Collocator to Collocator Interconnect Arrangement. If the Company provides the cables and connectors and/or runs the cable, the applicable cable and labor rates in Section 19.16 will be applied.

(J) **Optical Facility Terminations**

CLEC's who request access to unbundled dark fiber and unbundled optical interoffice facilities may apply for a fiber optic patchcord connection(s) between the Company's fiber distribution panel (FDP) and the CLEC's collocated transmission equipment and facilities. The fiber optic patchcord cross connect is limited in use solely in conjunction with access to unbundled dark fiber, unbundled optical interoffice facilities, and Dedicated Transit Service.

(C)

(M)

(M) Material moved from Page 12.

Third Revised Page 13 Canceling Second Revised Page 13

19. COLLOCATION SERVICE

19.4 Installation and Operation (Continued)

19.4.4 Access to Collocation Space

The Company will permit CLEC employees, agents, and contractors approved by the Company to have direct access to CLEC caged or cageless collocated equipment twenty-four (24) hours a day, seven (7) days a week and (C) reasonable access to the Company's restroom facilities. CLEC employees, agents, or contractors must comply with the policies and practices of the Company pertaining to fire, safety, and security as described in the Company's Security Procedures and Requirements Guidelines, which are attached to the Collocation Services Packet. The Company reserves the right, with 24 hours prior notice to the CLEC, to access the CLEC's collocated partitioned space to perform periodic inspections to ensure compliance with Company installation, safety and security practices. Where the CLEC shares a common entrance to the premises with the Company, the reasonable use of shared building facilities, e.g., elevators, unrestricted corridors, etc., will be permitted. Verizon reserves the right to permanently remove and/or deny access from Company premises any CLEC employee, agent or contractor who violates the Company's policies, work rules, or business conduct standards, or otherwise poses a security risk to the Company.

19.4.5 Network Outage, Damage and Reporting

The CLEC shall be responsible for:

- Any damage or network outage occurring as a result of CLEC owned or designated termination equipment in Company premises;
- Providing trouble report status when requested;
- Providing a contact number that is readily accessible 24 hours a day, 7 days a week;
- Notifying the Company of significant outages which could impact or degrade the Company's switches and services and provide estimated clearing time for restoral; and
- Testing its equipment to identify and clear a trouble report when the trouble has been sectionalized (isolated) to a CLEC service.

The Company will make every effort to contact the CLEC in the event CLEC equipment disrupts the network. If the Company is unable to make contact with the CLEC, the Company shall temporarily disconnect the CLEC's service, as provided in Section 19.4.7.

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

EFFECTIVE: October 13, 2001 ISSUED: September 28, 2001

Second Revised Page 14
Canceling First Revised Page 14

19. COLLOCATION SERVICE

19.4 Installation and Operation (Continued)

19.4.6 Security Requirements

(A) Background Tests; Training

All employees, agents and contractors of the CLEC must meet certain minimum requirements as set forth in the Company's Collocation Service Packet. When the CLEC submits the collocation ASR for caged or cageless collocation, or as soon as reasonably practicable thereafter, the CLEC must submit to the Company's Security Department, for prior approval, the background investigation certification form included in the Collocation Service Packet for all employees, agents and contractors that will require access to Company premises. The CLEC must agree that its employees/vendors with access to Company premises shall at all times adhere to the rules of conduct established by the Company for the premises and Company personnel and vendors.

The Company reserves the right to make changes to such procedures and rules to preserve the integrity and operation of Company network or facilities or to comply with applicable laws and regulations. The Company will provide the CLEC with written notice of such changes. Where applicable, the Company will provide information to the CLEC on the specific type of security training required so its employees can complete such training.

(B) Security Standards

The Company will be solely responsible for determining the appropriate level of security in each premises. The Company reserves the right to deny access to Company buildings for any CLEC employee, agent or contractor who cannot meet the Company's established security standards. Employees, agents or contractors of the CLEC are required to meet the same security requirements and adhere to the same work rules that Company employees and contractors are required to follow. The Company also reserves the right to deny access to Company buildings for CLEC employees, agents and contractors for falsification of records, violation of fire, safety or security practices and policies or other just cause.

CLEC employees, agents or contractors who meet the Company's established security standards will be provided access to the CLEC's collocation equipment twenty-four (24) hours a day, seven (7) days a week (T) and reasonable access to the Company's restroom facilities. If CLEC employees, agents or contractors (T) request and are granted access to other areas of the Company's premises, a Company employee, agent or contractor may accompany and observe the CLEC employee(s), agent(s) or contractor(s) at no cost to the CLEC.

The Company may use reasonable security measures to protect its equipment, including, for example, enclosing its equipment in its own cage or other separation, utilizing monitored card reader systems, digital security cameras, badges with computerized tracking systems, identification swipe cards, keyed access and/or logs, as deemed appropriate by the Company.

The Company may require CLEC employees and contractors to use a central or separate entrance to the (N) Company's premises, provided, however, that where the Company requires that CLEC employees or contractors access collocated equipment only through a separate entrance, employees and contractors of the Company's affiliates and subsidiaries will be subject to the same restriction.

The Company may construct or require the construction of a separate entrance to access caged and cageless collocation space, provided that each of the following conditions is met: (i) Construction of a separate entrance is technically feasible; (ii) Either legitimate security concerns, or operational constraints unrelated to the incumbent's or any of its affiliates' or subsidiaries competitive concerns, warrant such separation; (iii) Construction of a separate entrance will not artificially delay collocation provisioning; and (iv) Construction of a separate entrance will not materially increase the CLEC's collocation costs.

(N)

EFFECTIVE: October 13, 2001

ISSUED: September 28, 2001

First Revised Page 15 Canceling Original Page 15

19. COLLOCATION SERVICE

19.4 Installation and Operation (Continued)

(M)

19.4.6 Security Requirements (Continued)

(C) Access Cards/Identification

Access cards or keys will be provided to no more than a reasonable number of CLEC appointed individuals for each Company premises. All CLEC employees, agents and contractors requesting access to the premises (C) are required to have a photo identification card, which identifies the person by name and the name of the CLEC. The ID must be worn on the individual's exterior clothing while on Company premises. The Company will provide the CLEC with instructions and necessary access cards or keys to obtain access to Company buildings.

The CLEC is required to immediately notify the Company by the most expeditious means, when any CLEC employee, agent or contractor with access privileges to Company buildings is no longer in its employ, or when keys, access cards or other means of obtaining access to Company buildings are lost, stolen or not returned by an employee, agent or contractor no longer in its employ. The CLEC is responsible for the immediate retrieval and return to the Company of all keys, access cards or other means of obtaining access to Company buildings upon termination of employment of an employee and/or termination of service. The CLEC shall be (C) responsible for the replacement cost of keys, access cards or other means of obtaining access when lost, stolen or failure of it or its employee, agent or contractor to return them to the Company.

(M)

Second Revised Page 16 Canceling First Revised Page 16

19. COLLOCATION SERVICE

19.4 Installation and Operation (Continued)

19.4.7 Emergency Access

The CLEC is responsible for providing a contact number that is readily accessible 24 hours a day, 7 days a week. The CLEC will provide access to its collocation space at all times to allow the Company to react to emergencies, to maintain the building operating systems (where applicable and necessary) and to ensure compliance with OSHA/Company regulations and standards related to fire, safety, health and environmental safeguards. The Company will attempt to notify the CLEC in advance of any such emergency access. If advance notification is not possible the Company will provide notification of any such entry to the CLEC as soon as possible following the entry, indicating the reasons for the entry and any actions taken which might impact CLEC facilities or equipment and its ability to provide service. The Company will restrict access to CLEC collocation space to persons necessary to handle such an emergency.

The emergency provisioning and restoration of interconnection service shall be in accordance with Part 64, Subpart D, Paragraph 64.401, of the FCC's Rules and Regulations, which specifies the priority for such activities. The Company reserves the right, without prior notice, to access CLEC collocation space in an emergency, such as fire or other unsafe conditions, or for purposes of averting any threat of harm imposed by the CLEC or CLEC equipment upon the operation of Company equipment, facilities and/or employees located outside the CLEC's collocation space. The Company will notify the CLEC as soon as possible when such an event has occurred.

In case of a Company work stoppage, CLEC employees, contractors or agents will comply with the emergency operation procedures established by the Company. Such emergency procedures should not directly affect CLEC access to its premises, or ability to provide service. The CLEC will notify the Company point of contact of any work stoppages by CLEC employees.

19.5 Space Requirements

19.5.1 Space Availability

The Company shall permit the CLEC to secure collocation space on a first-come, first-serve priority basis upon the Company's receipt of fifty percent (50%) of the applicable nonrecurring Charges (NRCs) described in Section 19.4.1. If the Company is unable to accommodate caged and cageless collocation requests at a premises due to space limitations or other technical reasons, the Company will post a list of all such sites on its Website and will update the list within ten (10) calendar days of any known changes. This information will be listed at the following public Internet URL:

http://www.gte.com/regulatory

(T)

(D)

EFFECTIVE: December 2, 2000 ISSUED: November 17, 2000

19.5 Space Requirements (Continued)

19.5.1 **Space Availability (Continued)**

Where the Company is unable to accommodate caged and cageless collocation requests at a premises due to space (N) limitations or other technical reasons, the Company shall submit a Notice of Intent to Seek Waiver of Physical Collocation Requirements to the state commission on the same date of its initial response to the CLEC. The Notice shall include a basic statement of the reason for the denial (technically infeasible or lack of space). If the CLEC requests a Collocation Space Report for the premises, as described in Section 19.5.5, the Company shall submit a copy of the Report to the state commission.

If the Company is unable to accommodate a caged and cageless collocation request at a premises due to space limitations, the Company shall allow the CLEC and state commission staff to tour, without charge, the entire premises within 10 calendar days of the tour request. If the CLEC believes there is no basis for denial of its physical collocation application, it shall promptly notify the Company. Within 20 calendar days of the tour, the CLEC and Company shall file reports outlining their tour findings in the State Commission docket established to address the Company's Petition for Waiver.

If the Company is unable to accommodate a caged and cageless collocation request at a premises due to space limitations or other technical reasons, the Company shall submit to the state commission and CLEC a Petition for Waiver of the Collocation Requirements within 20 calendar days of submitting its Notice of Intent to request a Waiver. If the Petition of Waiver is based on space limitations, the Petition shall include the following information which will be considered Confidential in accordance with Rule 25-22.006, Florida Administrative Code:

- (1) Central Office Language Identifier, where applicable.
- (2) Identity of the CLEC, including the amount of space sought.
- (3) Total amount of space at the premises.
- (4) Floor plans, including measurements of the Company's premises showing: (a) Space housing the Company's nonregulated network equipment space or administrative offices; (b) Space housing obsolete and unused equipment; (c) Space that does not currently house Company equipment or administrative offices but is reserved by the Company for future use, including the intended purpose of each area and the forecasted year of use; (d) Space occupied by CLECs for the purpose of network interconnection or access to unbundled network elements; (e) Space, if any, occupied by third parties for other purposes, including identification of the uses of such space; (f) Remaining space, if any; (g) Identification of switch turnaround plans and other equipment removal plans and timelines, if any; (h) Central Office rearrangement/expansion plans, if any; and (i) Description of other plans, if any, that may relieve space exhaustion.
- (5) Floor loading requirements.

The State Commission Staff shall prepare a recommendation for consideration by the state commission at an Agenda Conference to be held within 45 calendar days of the post-tour report. The staff recommendation shall address whether the Company's Petition for Waiver should be granted based upon the information provided by the Company and the CLEC and information gathered by the Commission staff on the tour of the central office premises. The Commission's decision on the Company's Petition shall be issued as a final agency action. If the Commission grants the Petition, the Company will not have to justify subsequent denials of space to other CLECs. The Company shall however, advise the CLECs and the Commission when there are material changes in the central office premises that could affect a collocation request.

The Company will accept a Letter of Intent from a CLEC to collocate in premises where a Waiver from Caged and Cageless Collocation has been granted and the CLEC has not previously applied for Cagel or Cageless collocation at the premises. If a CLEC submits a Letter of Intent, a non-refundable Minor Augment Fee will be charged to the CLEC. (N)

EFFECTIVE: July 8, 2000

Third Revised Page 18 **Canceling Second Revised Page 18**

19. COLLOCATION SERVICE

19.5 **Space Requirements** (Continued)

19.5.1 **Space Availability (Continued)**

For those Company premises that have been granted a Waiver from Caged and Cageless collocation, the Company will notify the State Commission and CLECs sixty (60) calendar days in advance of new Caged and Cageless collocation space becoming available. The Company will inform the State Commission and CLECs previously requesting collocation space at the premises of the new space availability in writing. In the event the Company's determination that collocation space will be available does not allow for sixty (60) calendar days' notice, the Company will notify the State Commission and requesting CLECs within two business days of this determination.

When Caged and Cageless collocation space becomes available in a premises where space was previously unavailable, CLECs who previously applied for collocation space or submitted a Letter of Intent will given priority for the new collocation space based on the date of their prior application or the date received of the Letter of Intent for the premises. If a CLEC has accepted Virtual collocation after being denied Caged or Cageless collocation space, the CLEC will still be given priority for the new collocation space based on the denial date of their prior Caged or Cageless application. Any collocation space remaining after the new space has been offered to prior applicants will be available on a first-come, first-served basis based on the receipt of 50% of the NRCs.

If the Company is unable to meet the time frames applicable to any of the requirements identified in this Section, the Company shall seek an extension of time from the Commission at least three business days prior to the expiration of the identified time frame.

19.5.2 Minimum/Maximum/Additional Space

The minimum amount of floor space available to each CLEC at the time of the initial application will be twenty-five (25) square feet of caged collocation space or one (1) single bay in the case of cageless collocation. The maximum amount of space available in a specific premises to each CLEC will be limited to the amount of existing suitable space which is technically feasible to support the collocation arrangement requested. Existing suitable space is defined as available space in a premises which does not require the addition of AC/DC power, heat and air conditioning, battery and/or generator back-up power and other requirements necessary for provisioning collocation services. The Company will not deny a CLEC's collocation request if vacant, unconditioned space is the only space available for collocation. The Company will modify the vacant, unconditioned space to suitable space in order to support the collocation arrangement requested. Additional space to provide for caged, cageless and/or adjacent collocation will be provided on a per request basis, where feasible, and where space is being efficiently used. Additional space can be requested by a CLEC by completing and submitting a new application form and the applicable non-refundable engineering fee set forth in Section 19.16. The Company will not be required to lease additional space when available (T) space has been exhausted.

19.5.3 Use of Space

The Company and CLEC will work cooperatively to determine proper space requirements, and efficient use of space. In addition to other applicable requirements set forth in this Tariff, the CLEC shall install all its equipment within its designated area in contiguous line-ups in order to optimize the utilization of space within Company premises. The CLEC shall use the collocation space solely for the purposes of installing, maintaining and operating its equipment to interconnect for the exchange of traffic with the Company and/or for purposes of accessing unbundled network elements. The CLEC shall not construct improvements or make alterations or repairs to the collocation space without the prior written approval of the Company. The collocation space may not be used for administrative purposes and may not be used as CLEC employee(s) work location, office or retail space, or storage. The collocation space shall not be used as the CLEC's mailing or shipping address.

Third Revised Page 19 Canceling Second Revised Page 19

19. COLLOCATION SERVICE

19.5 Space Requirements (Continued)

19.5.4 Reservation of Space

The Company reserves the right to manage its own premises conduit requirements and to reserve vacant space for planned facilities. The Company will retain and reserve a limited amount of vacant floor space within its premises for its own specific future uses on terms no more favorable than applicable to other CLECs seeking to reserve collocation space for their future use. If the remaining vacant floor space within a premises is reserved for the Company's own specific future use, the premises will be exempt from future caged and cageless collocation requests in accordance with the guidelines and procedures of Section 19.5.1. The CLEC shall not be permitted to reserve premises cable space or conduit system. If new conduit is required, the Company will negotiate with the CLEC to determine an alternative arrangement for the specific location. The CLEC will be allowed to reserve collocation space for its caged/cageless arrangements based on its documented forecast provided to the Company and subject to space availability. Such forecast must demonstrate a legitimate need to reserve the space for use on terms no more favorable than applicable to the Company seeking to reserve vacant space for its own specific use. CLECs shall provide the Company with two-year forecasts, on an annual basis, to assist the Company in premises planning. CLEC cageless collocation bays may not be used solely for the purpose of storing CLEC equipment.

19.5.5 Collocation Space Report

Upon request by the CLEC and upon its signing a collocation nondisclosure agreement, the Company will make available a Collocation Space Report with the following information for the premises requested:

Detailed description and amount of caged and cageless collocation space available;

(T)

- Number of telecommunications carriers with existing collocation arrangements;
- Modifications of the use of space since the last Collocation Space Report requested; and,
- Measures being taken, if any, to make additional Collocation spaces available.

The Collocation Space Report is not required prior to the submission of a collocation application for a specific premises in order to determine collocation space availability for the premises. The Collocation Space Report will be provided to a CLEC within ten (10) calendar days of the request, provided the request is submitted during the ordinary course of business. A Collocation Space Report fee, as specified in Section 19.16, will be assessed per request and per premises.

19.5.6 Reclamation

When initiating an application form, the CLEC must have started installing equipment approved for collocation at the Company premises within a reasonable period of time, not to exceed six (6) months from the date the collocation arrangement is accepted. If the CLEC does not utilize its collocation space within the established time period, and has not met the space reservation requirements of Section 19.5.4, the Company may reclaim the unused collocation space to accommodate another CLEC request or the Company's future space requirements.

The Company shall have the right, for good cause shown, and upon six (6) months' notice, to reclaim any collocation space, cable space or conduit space in order to fulfill its obligation under public service law and its tariffs to provide telecommunication services to its end users. In such cases, the Company will reimburse the CLEC for reasonable direct costs and expenses in connection with such reclamation. The Company will make every reasonable effort to find other alternatives before attempting to reclaim any such space.

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA EFFECTIVE: October 13, 2001 ISSUED: September 28, 2001

Second Revised Page 20 Canceling First Revised Page 20

19. COLLOCATION SERVICE

19.6 Liability and Indemnification



- (A) No liability shall attach to the Company for damages arising from errors, mistakes, omissions, interruptions, or delays of the Company, its agents, servants or employees, in the course of establishing, furnishing, rearranging, moving, terminating, or changing the service or facilities (including the obtaining or furnishing of information in respect thereof or with respect to the subscribers or users of the service or facilities) in the absence of gross negligence or willful misconduct.
 - Subject to the preceding and to the provisions following, with respect to any claim or suit, by a CLEC or by any others, for damages associated with the installation, provision, termination, maintenance, repair or restoration of service, the Company's liability, if any, shall not exceed an amount equal to the proportionate charge for the service by the Company for the service for the period during which service was affected.
- (B) The Company shall not be liable for any act or omission of any other party furnishing a portion of service used in connection with the services herein.
- (C) The Company is not liable for damages to the CLEC premises resulting from the furnishing of service, including the installation and removal of equipment and associated wiring, unless the damage is caused by the Company's gross negligence or willful misconduct.
- (D) The Company shall be indemnified, defended and held harmless by the CLEC and/or its end user against any claim, loss or damage arising from the use of services offered under this tariff, involving:
 - (1) Claims for libel, slander, invasion of privacy, or infringement of copyright arising from the material transmitted over the Company's facilities;
 - (2) Claims for patent infringement arising from the CLEC's or its end user's acts combining or using the service furnished by the Company in connection with facilities or equipment furnished by the end user or the CLEC;
 - (3) All other claims arising out of any act or omission of the end user and/or CLEC in the course of using services provided pursuant to this tariff;
 - (4) All claims, including but not limited to injuries to persons or property from voltages or currents, arising out of any act or omission of the CLEC or its end user in connection with facilities provided by the Company, the CLEC, or the end user; or
 - (5) The Company shall not be liable to the CLEC or its customers in connection with the provision or use of the services provided under this tariff for indirect, incidental, consequential, reliance or special damages, including (without limitation) damages for lost profits, regardless of the form of action, whether in contract, indemnity, warranty, strict liability, or tort, including (without limitation) negligence of any kind, even if the Company has been advised of the possibility of such loss or damage.

(N)

EFFECTIVE: July 12, 2001

19.6 <u>Liability and Indemnification (Continued)</u>

(N)

- (E) The Company does not guarantee or make any warranty with respect to its services when used in an explosive atmosphere. The Company shall be indemnified, defended and held harmless by the CLEC from any and all claims by any person relating to such CLEC's use of services so provided.
- (F) No license under patents (other than the limited license to use) is granted by the Company or shall be implied or arise by estoppel, with respect to any service offered under this tariff.
- (G) The Company's failure to provide or maintain services under this tariff shall be excused by labor difficulties, governmental orders, civil commotions, criminal actions taken against the Company, acts of God and other circumstances beyond the Company's reasonable control.
- (H) The Company shall not be liable for any act or omission of any other entity furnishing to the CLEC facilities, equipment, or services used in conjunction with the services provided under this tariff. Nor shall the Company be liable for any damages or losses due to unauthorized use of the services or the failure or negligence of the CLEC or CLEC end user, or due to the failure of equipment, facilities, or services provided by the CLEC or its end user.
- (I) Neither party shall be liable to the other or to any third party for any physical damage to each other's facilities or equipment within the central office, unless caused by the gross negligence or willful misconduct of the party's agents or employees.
- (J) The CLEC shall indemnify, defend and save hamnless the Company from and against any and all losses, claims, demands, causes of action and costs, including attorney's fees, whether suffered, made, instituted or asserted by the CLEC or by any other party or person for damages to property and injury or death to persons, including payments made under any worker's compensation law or under any plan for employees' disability and death benefits, which may arise out of or be caused by the installation, maintenance, repair, replacement, presence, use or removal of the CLEC's equipment or facilities or by their proximity to the equipment or facilities or all parties occupying space within or on the exterior of the Company's central office(s), or by any act or omission of the Company, its employees, agents, former or striking employees, or contractors, in connection therewith, unless caused by gross negligence or willful misconduct on the part of the Company. These provisions shall survive the termination, cancellation, modification or rescission of the tariff for at least 18 months from the date of the termination.

The Company shall indemnify, defend and save harmless the CLEC from and against any and all losses, claims, demands, causes of action and costs, including attorneys' fees, whether suffered, made, instituted or asserted by the Company or by any other party or person for damages to property and injury or death to persons, including payments made under any worker's compensation law or under any plan for employees' disability and death benefits, which may arise out of or be caused by the Company's provision of service within or on the exterior of the central office of by an act or omission of the CLEC, its employees, agents, former or striking employees, or contractors, in connection therewith, unless caused by gross negligence or willful misconduct on the part of the CLEC.

A Party's obligation to indemnify the other Party as provided herein shall be conditioned upon the following:

- (1) The indemnified Party shall promptly notify the indemnifying Party of any action taken against the indemnified Party relating to the indemnification. However, the failure to give such notice shall release the Indemnifying Party from its obligations under this Section only to the extent the failure to give such notice has prejudiced the indemnifying Party.
- (2) The indemnifying Party shall have sole authority to defend any such action, including the selection of legal counsel, and the indemnified Party may engage separate legal counsel only at the indemnified Party's sole cost and expense.

(N)

19. COLLOCATION SERVICE

19.6 <u>Liability and Indemnification (Continued)</u>

- (3) In no event shall the indemnifying Party settle or consent to any judgment in an action without the prior written consent of the indemnified Party, which consent shall not be unreasonably withheld. However, in the event the settlement or judgment requires a contribution from or affects the rights of the indemnified Party, the indemnified Party shall have the right to refuse such settlement or judgment and, at its own cost and expense, take over the defense against such Loss, provided that in such event the indemnifying Party shall not be responsible for, nor shall it be obligated to indemnify the indemnified Party against, the Loss for any amount in excess of such refused settlement or judgment.
- (4) The indemnified Party shall, in all cases, assert any and all provisions in its Tariffs that limit liability to third parties as a bar to any recovery by the third party claimant in excess of such limitation of liability. The indemnified Party shall offer the indemnifying Party all reasonable cooperation and assistance in the defense of any such action.
- (K) The CLEC shall indemnify, defend and save harmless the Company from and against any and all losses, claims, demands, causes of action, damages and costs, including but not limited to attorney's fees and damages costs, and expense of relocating conduit systems resulting from loss of right-of-way or property owner consents, which may arise out of or be caused by the presence, in, or the occupancy of the central office by the CLEC, and/or acts by the CLEC, its employees, agents or contractors.
- (L) The CLEC shall indemnify, defend, and hold harmless the Company, its directors, officers and employees, servants, agents, affiliates and parent, from and against any and all claims, cost, expense or liability of any kind, including but not limited to reasonable attorney's fees, arising out of or relating to CLEC installation and operation of its facilities or equipment within the multiplexing node, roof space and transmitter space.
- (M) The CLEC represents, warrants and covenants that it shall comply with all applicable federal, state or local law, ordinance, rule or regulations, including but not limited to, any applicable environmental, fire, OSHA or zoning laws. The CLEC shall indemnify, defend, and hold harmless the Company, its directors, officers and employees, servants, agents, affiliates and parent, from and against any and all claims, cost, expense or liability of any kind including but not limited to fines or penalties arising out of any breach of the foregoing by the CLEC, its directors, officers, employees, servants, agents, affiliates and parent. These provisions shall survive the termination, cancellation, modification or rescission of the tariff for at least 18 months from the date of the termination.
- (N) The Company represents, warrants and covenants that it shall comply with all applicable federal, state or local law, ordinance, rule or regulations, in connection with its provision of service within or on the exterior of the central office, including but not limited to, any applicable environmental, fire, OSHA or zoning laws. The Company shall indemnify, defend, and hold harmless the CLEC, its directors, officers, employees, agents or contractors, from and against any and all claims, cost, expense or liability of any kind including but not limited to fines or penalties arising out of any breach of the foregoing by the Company, its directors, officers and employees, servants, agents, affiliates and parent.
- (O) The Company and the CLEC shall each be responsible for all persons under their control or aegis working in compliance herewith, satisfactorily, and in harmony with all others working in or on the exterior of the central office and, as appropriate, cable space.

(N)

EFFECTIVE: July 12, 2001

First Revised Page 21 **Canceling Original Page 21**

19. COLLOCATION SERVICE

19.7 Insurance

(M)

- The CLEC shall, at its sole cost and expense, obtain, maintain, pay for and keep in force the following minimum (C) 19.7.1 insurance, underwritten by an insurance company(s) having an A. M. Best insurance rating of at least A-, financial size category VII.
- - (A) Commercial general liability coverage on an occurrence basis in an amount of \$1,000,000 combined single limit for bodily injury and property damage with a policy aggregate per location of \$2,000,000. This coverage shall include contractual liability.
 - (B) Umbrella/Excess Liability coverage in an amount of \$10,000,000 excess of coverage specified in (A) above.
 - All Risk Property coverage on a full replacement cost basis insuring all of the CLEC's real and personal (C) property located on or within Company premises. The CLEC may also elect to purchase business interruption and contingent business interruption insurance, knowing that the Company has no liability for loss of profit or revenues should an interruption of service occur.
 - (D) Statutory Workers Compensation coverage.
 - (E) Employers Liability coverage in an amount of \$500,000 each accident.
 - (F) Commercial Automobile Liability coverage insuring all owned, hired and non-owned automobiles.

(C) (M)

Notwithstanding anything herein to the contrary, the coverage requirements described in (C) through (F) above (N) shall only be required if the CLEC orders collocation services pursuant to this Tariff. The minimum amounts of insurance required in this section may be satisfied by the CLEC purchasing primary coverage in the amounts specified or by the CLEC buying a separate umbrella and/or excess policy together with lower limit primary underlying coverage. The structure of the coverage is at the CLEC's option, so long as the total amount of insurance meets the Company's requirements.

(N)

(C)

19.7.2 **Deductibles**

Any deductibles, self-insured retentions (SIR), lost limits, retentions, etc. (collectively, "retentions") must be (N)

disclosed on a certificate of insurance provided to the Company, and the Company reserves the right to reject any such retentions in its reasonable discretion. All retentions shall be the responsibility of the CLEC.

(N) (D)

(D) (C) (M)

19.7.3 Additional Insureds

> The Company and its affiliates (which includes any corporation controlled by, controlling or in common control with the Company parent corporation), its respective directors, officers and employees shall be named as additional insureds under all General Liability and Umbrella/Excess Liability Policies obtained by the CLEC. endorsement shall provide that such additional insurance is primary insurance and shall not contribute with any insurance or self-insurance that the Company has secured to protect itself. All of the insurance afforded by the CLEC shall be primary in all respects, including the CLEC's Umbrella/Excess Liability insurance. The Company's insurance coverage shall be excess over any indemnification and insurance afforded by the CLEC and required hereby.

(C) (M)

Certain material appearing on this page formerly appeared on Original Page 12.

(N)

Second Revised Page 22 Canceling First Revised Page 22

19. COLLOCATION SERVICE

19.7 <u>insurance</u> (Continued)

19.7.4 Waiver of Subrogation Rights

The CLEC waives and will require all of its insurers to waive all rights of subrogation against the Company (including the Company parent Corporation and any other affiliated and/or managed entity), its directors, officers and employees, agents or assigns, whether in contract, tort (including negligence and strict liability) or otherwise.

19.7.5 Evidence of Insurance

All insurance must be in effect on or before the Company authorizes access by CLEC employees or placement of CLEC equipment or facilities within the Company's premises and such insurance shall remain in force as long as the CLEC's facilities remain within any space governed by this Tariff. If the CLEC fails to maintain the coverage, the Company may pay the premiums and seek reimbursement from the CLEC. Failure to make a timely reimbursement will result in disconnection of service. The CLEC agrees to submit to the Company a certificate of insurance ACORD Form 25-S (1/95), or latest edition, such certificate to be signed by a duly authorized officer or agent of the Insurer, certifying that the minimum insurance coverages and conditions set forth herein are in effect, and that the Company will receive at least thirty (30) days notice of policy cancellation, expiration or non-renewal.

At least thirty (30) days prior to the expiration of the policy, the Company must be furnished satisfactory evidence that such policy has been or will be renewed or replaced by another policy. At the Company's request, the CLEC shall provide copies of the insurance provisions or endorsements as evidence that the required insurance has been procured, and that the Company has been named as an additional insured, prior to commencement of any service. In no event shall permitting CLEC access be construed as a waiver of the right of the Company to assert a claim against the CLEC for breach of the obligations established in this section.

19.7.6 Compliance Requirements

The CLEC shall require its contractors to comply with each of the provisions of this insurance section. This includes, but is not limited to, maintaining the minimum insurance coverages and limits, naming the Company (including the Company's parent corporation and any other affiliated and/or managed entity) as an additional insured under all liability insurance policies, and waiving all rights of subrogation against the Company (including the Company's parent Corporation and any other affiliated and/or managed entity), its directors, officers and employees, agents or assigns, whether in contract, tort (including negligence and strict liability) or otherwise. Prior to commencement of any work, the CLEC shall require and maintain certificates of insurance from each contractor evidencing the required coverages. At the Company's request, the CLEC shall supply to the Company copies of such certificates of insurance or require the contractors to provide insurance provisions or endorsements as evidence that the required insurance has been procured. The CLEC must also conform to the recommendation(s) made by the Company's fire insurance company, which the Company has already agreed to or shall hereafter agree to.

19.7.7 Self Insurance

If the CLEC net worth exceeds \$100,000,000, the CLEC may elect to self insure and thereby assume the coverages, protections and payments that otherwise would have been provided or made to or on behalf of the Company under the insurance provisions set forth in this section. If the CLEC self insures, the CLEC shall furnish to the Company, and keep current, evidence of such net worth that is attested to by one of the corporate officers.

The CLEC is subject to the same liability and indemnification provisions set forth herein.

(N)

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA EFFECTIVE: July 12, 2001 ISSUED: June 27, 2001

19.8 Confidentiality

(M)

In addition to its other confidentiality obligations hereunder, the CLEC shall not use or disclose and shall hold in confidence all information of a competitive nature provided to it by the Company in connection with Collocation, or known to a CLEC as a result of its access to Company premises, or as a result of the interconnection of its (C) equipment to Company facilities. Similarly, the Company shall not use or disclose and shall hold in confidence all information of a competitive nature provided to it by a CLEC in connection with Collocation, or known to the Company as a result of the interconnection of the CLEC's equipment to Company facilities. Such information is to be considered proprietary and shared within the Company and the CLEC on a need to know basis only. Neither the Company nor the CLEC shall be obligated to hold in confidence information that:

- Was already known to the CLEC free of any obligation to keep such information confidential;
- Was or becomes publicly available by other than unauthorized disclosure; or
- Was rightfully obtained from a third party not obligated to hold such information in confidence.

(M)

Certain material appearing on this page formerly appeared on Original Page 13.

19.9 Casualty

(M)

If the collocation equipment location in the Company premises is rendered wholly unusable through no fault of the (C) CLEC, or if the building shall be so damaged that the Company shall decide to demolish it, rebuild it, or abandon it for premises purposes (whether or not the demised premises are damaged in whole or in part), then, in any of such (C) events, the Company may elect to terminate the collocation arrangements in the damaged building by providing written notification to CLECs as soon as practicable but no later than one hundred eighty (180) days after such casualty, specifying a date for the termination of the Collocation arrangements. The termination date shall not be more than sixty (60) days after the giving of such notice. Upon the date specified in such notice, the term of the collocation arrangement shall expire as fully and completely as if such date were the date set forth for the termination of the arrangement. CLECs shall immediately quit, surrender and vacate the premises without prejudice. Unless the Company serves a termination notice as provided for herein, it shall make the repairs and restorations with all reasonable expedition, subject to delays due to adjustment of insurance claims, labor troubles and causes beyond the Company's reasonable control. After any such casualty, CLECs shall cooperate with the Company's restoration by removing from the collocation space, as promptly as reasonably possible, all of their salvageable inventory and movable equipment, furniture and other property. The Company will work cooperatively with the CLECs to minimize any disruption to service, resulting from any damage. The Company shall provide written notification to CLECs detailing its plans to rebuild and will restore service as soon as practicable. In the event of termination, the Company's rights and remedies against CLECs in effect prior to such termination, and any fees owing, shall be paid up to such date. Any advance payments of fees made by CLECs for periods after such date, shall be returned.

(M)

Certain material appearing on this page formerly appeared on Original Page 13.

Second Revised Page 25 Cancels First Revised Page 25

EFFECTIVE: February 15, 2002

ISSUED: January 31, 2002

19. COLLOCATION SERVICE

19.10 Termination of Service

19.10.1 Grounds for Termination

Failure by the CLEC to comply with the terms and conditions of this Tariff, including nonpayment of rates and (T) charges, may result in termination of service. (T)

(T) (D)

(D)

Collocation arrangements will automatically terminate if the premises in the which the collocation space is located is closed, decommissioned or sold and no longer houses the Company's network facilities. At least one hundred eighty (180) days written notice will be given to the CLEC of events which may lead to the automatic termination of any such arrangement pursuant to this tariff, except when extraordinary circumstances require a shorter interval. In such cases, the Company will provide notice to the CLEC as soon as practicable. The Company will work with the CLEC to identify alternate collocation arrangements. The Company will work cooperatively with the CLEC to minimize any potential for service interruption resulting from such actions.

In addition to the other grounds for termination of collocation services set forth herein, the Company also reserves the right to terminate such services upon thirty (30) days notice in the event the CLEC is not in conformance with Company standards and requirements, and/or imposes continued disruption and threat of harm to Company employees and/or network, or the Company's ability to provide service to other CLECs.

Second Revised Page 26 Canceling First Revised Page 26

19. COLLOCATION SERVICE

19.10 Termination of Service (Continued)

19.10.2 Effects of Termination

A CLEC must provide a minimum of thirty (30) calendar days written notice if the CLEC elects to terminate a existing collocation arrangement after acceptance of the collocation space. All monthly recurring charges will continue for thirty (30) calendar days from the date of the termination notice, or until the collocation space is vacated, whichever is longer.

If a CLEC has paid a non-recurring charge(s) for an asset in a collocation arrangement and is succeeded by another CLEC who uses the same asset, the initial CLEC will be credited the remaining undepreciated amount of the asset upon occupancy by the subsequent CLEC. The subsequent CLEC will be responsible for paying the remaining undepreciated amount of the cost. If the Company uses an asset for which a CLEC paid a non-recurring charge, the Company will make a pro rata refund to the CLEC. For purposes of calculating prorated refunds to the CLEC, the Company will use the economic life of the asset.

Upon the termination of collocation service, the CLEC shall disconnect and remove its equipment from the designated collocation space. Due to physical and technical constraints, removal of CLEC entrance facility cable will be at the Company's option. The Company reserves the right to remove CLEC equipment if the CLEC fails to remove and dispose (T) of the equipment within the thirty (30) days of discontinuance. The CLEC will be charged the appropriate additional labor charge(s) set forth in Section 19.16 for the removal of such equipment. Upon removal by the CLEC of all its equipment from the collocation space, it will reimburse the Company for the cost to restore the collocation space to its original condition at time of occupancy. The cost will be applied based on the additional labor charges rate set forth in Section 19.16.

19.10.3 Cancellations and Acceptance Delays

If a CLEC elects to cancel a request for collocation when construction is in progress and prior to acceptance of the collocation space, the CLEC must do so in writing. Engineering/Major Augment fees submitted with the application and all other non-recurring charges for the Company's costs incurred in providing the collocation arrangement will not be (T) refunded. No monthly recurring charges will be billed to the CLEC.

If a CLEC elects to not accept a completed collocation arrangement, the CLEC must provide written notice within 30 calendar days of the scheduled completion date to avoid incurring any monthly recurring charges. Engineering/Major Augment fees submitted with the application and all other non-recurring charges for the Company's costs incurred in (T) providing the collocation arrangement will not be refunded.

19.11 Miscellaneous

The Company retains ownership of premises floor space, adjacent land and equipment used to provide all forms of collocation. The Company reserves for itself and its successors and assignees, the right to utilize the premises space in such a manner as will best enable it to fulfill its service requirements. The CLEC does not receive, as a result of entering into a collocation arrangement, any right, title or interest in Company wire center facility, the multiplexing node, multiplexing node enclosure, cable, cable space, cable racking, vault space or conduit space other than as expressly provided herein. To the extent that a CLEC requires use of a Company local exchange line, it must order a business local exchange access line (B1). A CLEC may not use Company official lines.

(T)

(D) (D)

(N)

EFFECTIVE: February 15, 2002 ISSUED: January 31, 2002

First Revised Page 26.1 Canceling Original Page 26.1

19. COLLOCATION SERVICE

19.12 Virtual Collocation

(A) Description

Under virtual collocation, the Company installs and maintains CLEC provided equipment, which is dedicated to the exclusive use of the CLEC in a collocation arrangement. A CLEC provides fiber-optic facilities through Company entrance manholes for connection to the CLEC virtually collocated transmission equipment that provides interconnection to Company facilities located in the premises.

The physical point of interface for connection to the virtual arrangement is referred to as manhole zero. From this manhole into the premises, the Company shall assume ownership of and maintain the fiber. From this manhole toward the CLEC's location, the fiber optic cable remains the CLEC's responsibility, with the CLEC performing all servicing and maintaining full ownership. If the CLEC is purchasing Company provided unbundled interoffice facilities as transport, the CLEC entrance fiber is not required. All elements/services shall be connected to the output cables of the virtual collocation arrangement using Company designated cable assignments, not channel assignments.

Virtual collocation is offered on a first come, first served basis and is provided subject to the availability of space and facilities in each premises where virtual collocation is requested.

Implementation Intervals and Planning

The Company and the CLEC shall work cooperatively to jointly plan the implementation milestones. The Company and the CLEC shall work cooperatively in meeting those milestones and deliverables as determined during the joint planning process. A preliminary schedule will be developed outlining major milestones including anticipated delivery dates for the CLEC-provided transmission equipment and for training.

The Company will notify the CLEC of issues or unanticipated delays, as they become known. The Company and the CLEC shall conduct additional joint planning meetings, as reasonably required, to ensure all known issues are discussed and to address any that may impact the implementation process. Planning meetings shall include establishment of schedule, identification of tests to be performed, spare plug-in/card requirements, test equipment, and determination of the final implementation schedule.

Application and price quote intervals for Virtual collocation requests are described in Section 19.3. For new virtual arrangements, the Company shall complete installation of standard Virtual collocation requests within sixty (60) calendar days of receipt of the application date, assuming that the material shipment, training, and construction intervals for the (C) improvements required to accommodate the request (e.g., HVAC, system/power plant upgrade/cables) are met. The Company and the CLEC shall work cooperatively to schedule each site on a priority-based order. The Company and the CLEC shall mutually agree upon intervals for non-standard arrangements. If the Company is unable to accommodate the CLEC's Virtual collocation request within the applicable number of days and the CLEC does not agree to an extension, the Company shall submit a Motion for Extension of Time to the Commission and CLEC within 45 calendar days of the application date. The Motion shall explain, in detail, the reasons necessitating the extension.

For augments to existing Virtual arrangements, the Company shall complete the augment request within sixty (60) calendar days of the application date, assuming that the material shipment, training and construction intervals for the improvements required to accommodate the request (e.g., HVAC, system/power plant upgrade/ cables) are met. If the Company is unable to accommodate the CLEC's Virtual augment request within the applicable number of days and the CLEC does not agree to an extension, the Company shall submit a Motion for Extension of Time to the Commission and CLEC within 45 calendar (C) days of the application date. The Motion shall explain, in detail, the reasons necessitating the extension.

(C)

(C)

(C)

EFFECTIVE: October 13, 2001

ISSUED: September 28, 2001

19.12 <u>Virtual Collocation (Continued)</u>

(N)

(B) Implementation Intervals and Planning (Continued)

The Company will request Virtual collocation forecasts from the CLEC on an annual basis, with each forecast covering a two-year period. The CLEC will be required to update the near-term (6-month) forecasted application dates. Information requested will include central office, month applications are expected to be sent, requested in-service month, and square footage required.

Unforecasted demand will be given a lesser priority than forecasted demand. The Company will make every attempt to meet standard intervals for unforecasted requests. However, if unanticipated requests push demand beyond the Company's capacity limits, the Company will negotiate longer intervals as required (and within reason). In general, if forecasts are received less than two (2) months prior to the application date, delivery of the floor space to the CLEC may be postponed by no more than sixty (60) calendar days.

(C) <u>Transmission Failure</u>

In the event of a transmission failure, the obligation to determine fault location, regardless of whether the fiber span is equipped with optical regeneration equipment, lies with the transmitting end. It is the responsibility of the receiving end to report incoming signal loss to the transmitting end.

(D) Accommodations

Upon receipt of a completed application and associated Virtual Engineering fee, the Company will conduct an application review, engineering review and site survey at the requested premises. The Company will notify the CLEC within ten business days of the results of this review and site survey.

The dedicated terminal equipment inside the Company's premises shall be provided by the CLEC and leased to the Company for the sum of one dollar after successful installation and equipment testing by the Company. The term of the operating lease will run for the duration of the virtual collocation arrangement, at which time the CLEC will remove the equipment. The CLEC will retain ownership of this equipment inside the premises. The Company will operate and maintain exclusive control over this equipment inside the premises.

Where the Company uses approved contractors for installation, maintenance or repair of Virtual collocation arrangements, the CLEC may hire the same approved contractors directly for installation, maintenance or repair of CLEC designated equipment.

Where the Company does not use contractors, CLEC designated equipment and CLEC provided facilities used in the provision of Virtual collocation will be installed, maintained and repaired by the Company. The Company will maintain and repair the CLEC designated equipment under the same timeframe and standards as its own equipment.

CLEC personnel are not allowed on the Company premises to maintain and repair on Virtual collocation equipment.

The Company shall monitor local premise and environmental alarms to support the equipment. The Company will notify the CLEC if a local office alarm detects an equipment affecting condition.

The Company will be responsible to pull the fiber into and through the cable entrance facility (i.e., vault) to the virtual collocation arrangement. All installations into the cable entrance facility are performed by Company personnel or its agents.

No virtual collocation arrangement will be placed in service by the Company until necessary training has been completed (refer to Section 19.12(K)).

(N)

EFFECTIVE: July 12, 2001

19. COLLOCATION SERVICE

19.12 <u>Virtual Collocation (Continued)</u>

(E) Plug-Ins and Spare Cards

When a plug-in/card is determined by the Company to be defective, the Company will label the plug-in as defective and place it in the CLEC-dedicated plug-in/card storage cabinet. The CLEC will be notified as the plug-in/card is replaced.

The Company will not provide spare plug-ins/cards under any circumstances, nor is the Company responsible for the CLEC's failure to replace defective plug-ins/cards. The Company shall not be held responsible if the CLEC provides an inadequate supply of plug-ins/cards. The Company will segregate and secure the CLEC-provided maintenance spares in the CLEC-provided spare plug-in/card cabinet.

The CLEC shall provide the shop-wired piece of equipment fully pre-equipped with working plug-ins/cards. In addition, the CLEC shall provide the Company with maintenance spares for each plug-in/card type. The number of maintenance spares shall be the manufacturer's recommended amount, unless otherwise mutually agreed by the Company and the CLEC, provided however, that in no event shall the number of spare plug-ins/cards be less than two of each type. These spares must be tested by the CLEC prior to delivery to the Company.

In addition to maintenance spares, the CLEC will also provide any unique tools or test equipment required to maintain, turn-up, or repair the equipment.

Upon receiving notification from the Company that a plug-in/card has been replaced, the CLEC is then responsible to contact the Company operations manager to arrange exchange and replacement of the plug-in/card. Exchanged, pretested spares shall be provided within one week of replacement of a defective plug-in/card.

Subject to premise space availability, the CLEC shall have the option of providing a stand-alone spare plug-in/card cabinet(s) or a rack-mountable spare plug-in/card cabinet(s), to the Company's specification, to house the spare plug-ins/cards. The spare plug-in/card cabinet(s) and minimum number of maintenance spares must be provided before the virtual collocation arrangement is completed and service is established.

The amount of spare plug-ins/cards required will be based on the manufacturer's recommended amount, unless otherwise mutually agreed by the Company and the CLEC.

(F) Safety and Technical Standards

The Company reserves all rights to terminate, modify or reconfigure the provision of service to the CLEC if, in the discretion of the Company, provision of service to the CLEC may in any way interfere with or adversely affect the Company's network or its ability to service other CLECs.

All CLEC equipment to be installed in Company premises must fully comply with the GR – 000063 – CORE, GR – 1089 – CORE and the Company's premise environmental and transmission standards in effect at the time of equipment installation. The equipment must also comply with the requirements in NIP 74165, as they relate to fire, safety, health, environmental, and network safeguards.

It is the CLEC's responsibility to demonstrate and provide to the Company adequate documentation from an accredited source certifying compliance. CLEC equipment must conform to the same specific risk/safety/hazard standards which the Company imposes on its own premises equipment as defined in RNSA – NEB – 95 – 0003, Revision 10 or higher.

CLEC equipment is not required to meet the same performance and reliability standards as the Company imposes on its own equipment as defined in RNSA – NEB - 95 - 0003, Revision 10 or higher. The CLEC may install equipment that has been deployed by the Company for five years or more with a proven safety record.

(N)

EFFECTIVE: July 12, 2001

FACILITIES FOR INTRASTATE ACCESS

19. COLLOCATION SERVICE

19.12 <u>Virtual Collocation (Continued)</u>

(F) Safety and Technical Standards (Continued)

All the CLEC's entrance facilities and splices must comply with TR – TSY – 00020, TR – NWT – 001058, BR – 760 – 200 – 030 and SR – TAP – 001421 as they relate to fire, safety, health, environmental safeguards and interference with the Company's services and facilities. Such requirements include, but are not limited to the following: (1) The fibers must be single mode; (2) The fiber optic units must be of loose tube (12 fibers) or ribbon (12 fibers) design; (3) The fiber cable must be marked according to the cable marking requirements in GR – 20 – CORE, Section 6.2.1 – 4; (4) The fiber must be identified according to the fiber and unit identification (color codes) in GR – 20 – CORE, Section 6.2.5; (5) Unless otherwise mutually agreed, the outer cable jacket shall consist of a polyethylene resin, carbon black, and suitable antioxidant system; and (6) Silica fibers shall be fusible with a commercially available fusion splicer(s) that is commonly used for this operation.

(G) Control Over Premises Based Equipment

The Company exercises exclusive physical control over the premises-based transmission equipment that terminates the CLEC's circuits and provides the installation, maintenance, and repair services necessary to assure proper operation of the virtually collocated facilities and equipment. Such work will be performed by the Company under the direction of the CLEC.

(H) Removal of Equipment

The Company reserves the right to remove facilities and equipment from its list of approved products if such products, facilities and equipment are determined to be no longer compliant with NEBS standards or GR – 1089 – CORE.

(I) Installation and Trouble Resolution

The Company will process and prioritize the trouble ticket in the same manner it does for its own equipment, including the dispatch of a technician to the equipment. The technician will contact the CLEC at the number provided and service the equipment as instructed and directed by the CLEC.

(N)

EFFECTIVE: July 12, 2001 ISSUED: June 27, 2001

19. COLLOCATION SERVICE

19.12 <u>Virtual Collocation (Continued)</u>

(J) Placement, Removal and Monitoring of Facilities and Equipment

From manhole zero toward the CLEC's location the fiber optic cable remains the CLEC's responsibility, with the CLEC performing all servicing and maintaining full ownership. The CLEC has the responsibility to remotely monitor and control their circuits terminating in the Company's premises, however, the CLEC will not enter the Company's premises under virtual collocation arrangements.

Performance and surveillance monitoring and trouble isolation shall be provided by the CLEC. A clear distinction must be made by the CLEC when submitting reports of troubles on the Company services/elements connected to the virtually collocated equipment and reports of troubles with the collocated equipment. The former can be handled using Company technicians and standard processes. The latter will require specially trained technicians familiar with the collocated equipment (refer to Section 19.12(K)).

When the CLEC isolates a trouble and determines that a Company technician should be dispatched to the equipment location for a servicing procedure, the CLEC shall enter a trouble ticket with the Company. The CLEC shall provide standard trouble information, including the virtual collocation arrangement's circuit identification, nature of the activity request, and the name and telephone number of the CLEC's technician/contact.

Responses to all equipment servicing needs will be at the CLEC's direction. Maintenance will not be performed without the CLEC's direct instruction and authorization.

If the CLEC is providing its own transport fiber for the virtual collocation arrangement, the CLEC will arrange placement of the fiber into manhole zero with enough length (as designated by the Company) to reach the virtual collocation arrangement.

Maintenance activity (trouble in the equipment) is to be tested, isolated and evaluated by the CLEC. Company technicians will perform the instructed activities on the equipment as specifically directed by the CLEC.

The CLEC shall provide, own, and operate the terminal equipment at their site outside the Company's premises.

(N)

EFFECTIVE: July 12, 2001

19.12 Virtual Collocation (Continued)

(N)

(K) Use of Non-Standard Equipment

When a CLEC requests a virtual collocation arrangement consisting of equipment which the Company does not use in its network nor has deployed in that particular premise to provide service to itself or another CLEC, the CLEC shall be responsible for training 50%, but no fewer than five, of the Company technicians in the administrative work unit responsible for servicing the equipment. Any special tools or electronic test sets that the Company does not have at the premises involved must be provided by the CLEC with adequate manufacturer's training.

The CLEC is responsible to arrange and pay all costs (including but not limited to transportation and lodging for Company technicians) to have Telephone Company technicians professionally trained by appropriate trainers certified on the specific equipment to be used to provide the virtual collocation arrangement to the CLEC. The CLEC shall also pay for the Company technicians' time subject to tariffed rates. When travel is required, travel expenses associated with training will be charged to the CLEC based on ticket stubs and/or receipts. This includes paying for mileage according to the IRS rates for personal car mileage or airfare, as appropriate. The CLEC also has the option of arranging and paying for all travel expenses for Company technicians directly.

In the event of an equipment upgrade, the CLEC must provide secondary training subject to the provisions contained herein.

(L) Additions and Rearrangements

Once the CLEC has established a virtual collocation arrangement, changes to the existing configuration, (including but not limited to, growing, upgrading, and/or reconfiguring the current equipment) are considered rearrangements to that virtual collocation arrangement. If the CLEC decides to rearrange an existing virtual collocation arrangement, the CLEC must submit a new application outlining the details of the rearrangement along with a Virtual Engineering/Major Augment fee.

(N)

Original Page 26.7

19. COLLOCATION SERVICE

19.12 Virtual Collocation (Continued)

(N)

(M) Application of Rates and Charges

(1) Billing

The Company will apply charges (e.g., nonrecurring and recurring rates for entry fiber, power, etc.) and commence billing for the virtual collocation arrangement upon completion of the installation, when it shall have finished all elements of the installation under its control. The readiness of the CLEC to utilize the completed virtual collocation arrangement will not impair the right of the Company to commence billing.

The Company shall charge the CLEC for all costs incurred in providing the virtual collocation arrangement, including, but not limited to, the Company's planning, engineering and installation time and costs incurred by the Company for inventory services. Any and all expenses associated with placing the CLEC's fiber in manhole zero, including license fees, shall be the responsibility of the CLEC.

(2) Virtual Engineering Fee

The Company will require a Virtual Engineering/Major Augment fee (NRC) per virtual collocation request, per premise or other Company location where the CLEC requests to establish virtual collocation. A Virtual Engineering/Major Augment fee is required to be submitted by the CLEC with its application. This fee applies for all new virtual collocation arrangements as well as subsequent additions to an existing arrangement, and provides for application processing, and for the Company's performance of an initial site visit and an engineering evaluation.

If the CLEC cancels or withdraws its request for a virtual collocation arrangement prior to turn-up, the CLEC will be liable for all costs and liabilities incurred by the Company in the developing, establishing, or otherwise furnishing the virtual collocation arrangement up to the point of cancellation or withdrawal.

(3) Other Virtual Collocation Rate Elements

The application, description, and rates of other Virtual Collocation rate elements are described in Sections 19.15 and 19.16.

(N)

19. COLLOCATION SERVICE

19.13 Microwave Collocation

(N)

Microwave collocation is available on a first-come first-served basis where technically feasible. The microwave equipment may include microwave antenna(s), mounts, towers or other antenna support equipment on the exterior of the building, and radio transmitter/receiver equipment located either inside or on the exterior of the building. All microwave antennas must be physically interconnected to Company facilities through the collocation arrangement.

(A) Accommodations

The Company will provide space within the cable riser, cable rack support structures and between the transmitter/receiver space and the roof space needed to reach the physical or virtual collocation arrangement and to access the Company's interconnection point. Waveguide may not be placed in Company cable risers or racks. The Company reserves the right to prohibit the installation of waveguide, metallic conduit and coaxial cable through or near sensitive equipment areas. The route of the waveguide and/or coaxial cable as well as any protection required will be discussed during the pre-construction survey.

The Company will designate the space in, on or above the exterior walls and roof of the premises which will constitute the roof space or transmitter/receiver space. The Company may require the CLEC's transmitter/receiver equipment to be installed in a locked cabinet which may be free standing, wall mounted or relay rack mounted. The Company may enclose the CLEC's multiplexing node or transmitter/receiver equipment in a cage or room.

At the option of the Company, the antenna support structure shall be built, owned and maintained by either the Company or by the CLEC. The Company reserves the right to use existing support structures for a CLEC's antenna, subject to space and capacity limitations. The Company also reserves the right to use any unused portion of a support structure owned by a CLEC for any reason, subject to the provisions set forth below. It shall be the responsibility of the owner of the support structure to maintain a record of the net book value of the structure. When the Company is the owner of the structure, it shall keep such records in accordance with the FCC's Part 32 uniform system of accounts. When the CLEC is the owner of the structure, it shall keep such records in accordance with generally accepted accounting principles.

The owner of the support structure shall use reasonable efforts to accommodate requests by other CLECs to use the support structure for microwave interconnection on a first-come first-served basis.

For those interconnecting via microwave facilities, transmitter/receiver equipment may be located in the CLEC's interior collocation space, or in a separate location inside or on the exterior of the building as determined by the Company.

(B) Security

The Company will permit the CLEC's employees, agents and contractors approved by the Company to have access to the areas where the CLEC's microwave antenna and associated equipment (e.g., tower and support structure, transmitter/receiver equipment, and waveguide and/or coaxial cable) is located during normal business hours for installation and routine maintenance, provided that the CLEC employees, agents and contractors comply with the policies and practices of the Company pertaining to fire, safety and security. Such approval will not be unreasonably withheld. During non-business hours, the Company will provide access on a per event basis.

The Company will also permit all approved employees, agents and contractors of the CLEC to have access to the CLEC's cable and associated equipment (e.g., repeaters). This will include access to riser cable, cableways, and any room or area necessary for access.

(N)

EFFECTIVE: July 12, 2001

19. COLLOCATION SERVICE

19.13 Microwave Collocation (Continued)

(C) Safety and Technical Standards

The Company reserves the right to remove facilities and equipment from its list of approved products if such products, facilities and equipment are determined to be no longer compliant with NEBS standards or electromagnetic compatibility and electrical safety generic criteria for network telecommunication equipment specified in GR – 1089 – CORE. The Company will provide 90 days notice of the change unless it is due to an emergency which renders notice impossible.

The Company reserves the right to review wind or ice loadings, etc., for antennas over 18 inches in diameter or for any multiple antenna installations, and to require changes necessary to insure that such loadings meet generally accepted engineering criteria for radio tower structures.

The minimum height of equipment placement, such as microwave antennas, must be eight feet from the roof. For masts, towers and/or antennas over 10 feet in height, the CLEC or if applicable, the Company, shall have the complete structure, including guys and supports, inspected every two years by an acceptable licensed professional engineer of its choice specializing in this type of inspection. For CLEC owned structures that are solely for the use of one CLEC's antenna(s), such inspection will be at the CLEC's own cost and expense. For structures used by multiple CLECs, the costs associated with such inspection shall be apportioned based on relative capacity ratios. A copy of this report may be filed with the Company within 10 days of the inspection. The owner shall be responsible to complete all maintenance and/or repairs, as recommended by the engineer, within 90 days.

The CLEC shall provide written notice to the Company of any complaint (and resolution of such complaint) by any governmental authority or others pertaining to the installation, maintenance or operation of the CLEC's facilities or equipment located in roof space or transmitter/receiver space. The CLEC also agrees to take all necessary corrective action.

All CLEC microwave equipment to be installed in or on the exterior of the Company premises must be on the Company's list of approved products, or equipment that is demonstrated as complying with the technical specifications described herein. Where a difference may exist in the specifications, the more stringent shall apply.

The CLEC must comply with Company technical specifications for microwave collocation interconnection specified in NIP – 74171 and the Company's digital switch environmental requirements specified in NIP – 74165, as they relate to fire, safety, health, environmental, and network safeguards, and ensure that CLEC provided equipment and installation activities do not act as a hindrance to Company services or facilities. The CLEC's equipment placed in or on roof space or transmitter/receiver space must also comply with all applicable rules and regulations of the FCC and the FAA.

CLEC facilities shall be placed, maintained, relocated or removed in accordance with the applicable requirements and specifications of the current edition of NIP – 74171, national electric code, the national electrical safety code, rules and regulations of the OSHA, and any governing authority having jurisdiction.

All CLEC microwave facilities must comply with Bellcore specifications regarding microwave and radio based transmission and equipment, CEF, BR -760-200-030, and SR -TAP-001421; and the Company's practices as they relate to fire, safety, health, environmental safeguards transmission and electrical grounding requirements, or interference with Company services or facilities.

The equipment located in, on or above the exterior walls or roof of the Company's building must either be on the Company's list of approved products or fully comply with requirements specified in GR – 63 – CORE, GR – 1089 – CORE and NIP 74171. This equipment must also comply with NIP – 74160, premise engineering environmental and transmission standards as they relate to fire, safety, health, environmental safeguards, or interference with Company service or facilities.

(N)

EFFECTIVE: July 12, 2001

19.13 Microwave Collocation (Continued)

(N)

(C) Safety and Technical Standards (Continued)

Each transmitter individually and all transmitters collectively at a given location shall comply with appropriate federal, state and/or local regulations governing the safe levels of radio frequency radiation. The minimum standard to be met by the CLEC in all cases is specified in ANSI C95.1 – 1982.

CLEC equipment must conform to the same specific risk, safety, hazard standards which the Company imposes on its own premises equipment as defined in RNSA – NEB – 95 – 0003, Revision 10 or higher. CLEC equipment is not required to meet the same performance and reliability standards as the Company imposes on its own equipment as defined in RNSA – NEB – 95 – 0003, Revision 10 or higher.

(D) Placement and Removal of Facilities and Equipment

Prior to installation of the CLEC's facilities or transmission equipment for microwave interconnection, the CLEC must obtain at its sole cost and expense all necessary licenses, permits, approvals, and/or variances for the installation and operation of the equipment and particular microwave system, and when applicable for any towers or support structures, as may be required by authorities having jurisdiction.

The CLEC is not permitted to penetrate the building exterior wall or roof when installing or maintaining transmission equipment and support structures. All building penetration will be done by the Company or a hired agent of the Company.

Any CLEC's equipment used to produce or extract moisture must be connected to existing or newly constructed building or roof top drainage systems, at the expense of the CLEC.

The CLEC will be responsible for supplying, installing, maintaining, repairing and servicing the following microwave specific equipment: Waveguide, waveguide conduit, and/or coaxial cable, the microwave antenna and associated tower and support structure and any associated equipment; and the transmitter/receiver equipment and any required grounding.

The CLEC may install equipment that has been deployed by the Company for five years or more with a proven safety record.

(E) Moves, Replacements or Other Modifications

Where the CLEC intends to modify, move, replace or add to equipment or facilities within or about the roof space or transmitter/receiver space(s) and requires special consideration (e.g., use of freight elevators, loading dock, staging area, etc.), the CLEC must request and receive written consent from the Company. Such consent will not be unreasonably withheld. The CLEC shall not make any changes from initial installation in terms of the number of transmitter/receivers, type of radio equipment, power output of transmitters or any other technical parameters without the prior written approval of the Company.

(N)

19. COLLOCATION SERVICE

19.13 Microwave Collocation (Continued)

(F) Space and Facilities

Monthly rates are applicable to each microwave CLEC for the space (generally on the premises roof) associated with Company or CLEC owned antenna support structures. The rate is calculated using the rate per square foot, multiplied by the square footage of the foot print, which resultant is multiplied by the CLEC's RCRs, (i.e., the sum of the RCRs of each of the CLEC's antennas).

Square footage for the footprint will be based on the length times width of the entire footprint formed on the horizontal plane (generally the roof top) by the antenna(s), tower(s), mount(s), guy wires and/or support structures used by the CLEC. For a non-rectangular footprint, the length will be measured at the longest part of the footprint and the width will be the widest part of the footprint.

The owner of the support structure may charge CLECs proposing to use the structure, on a one-time basis, for the following costs and/or values. Any incremental costs associated with installing the user's antenna, including but not limited to, the costs of engineering studies, roof penetrations, structural attachments, support structure modification or reinforcement, zoning and building permits. A portion of the net book value of the support structure based on the relative capacity ratio (RCR) of the user's proposed antenna(s) to be mounted on the structure. A user's RCR represents the percent of the total capacity of the support structure used by user's antenna(s) on the structure. Spare capacity shall be deemed to be that of the owner of the structure. RCRs shall be expressed as a two place decimal number, rounded to the nearest whole percent. The sum of all user's RCRs and the owner's RCR shall at all times equal 1.00. It shall be the responsibility of the owner of the structure to provide the proposed user the net book value of the structure at the time of the proposed use. Upon request, the owner shall also provide the proposed user accounting records or other documentation supporting the net book value.

The owner of the structure may not assess other users of the structure any charges in addition to the one-time charge described above, except that the owner of the structure may assess other user's a proportionate share of inspection costs and the Company may assess microwave CLECs monthly recurring charges for use of its roof space. At the time a CLEC (including the owner) proposes to attach additional antennas to an existing support structure, it shall be the responsibility of that CLEC to obtain, at their cost and expense, an engineering analysis by a registered structural engineer to determine the relative capacity ratio of all antennas on the structure, including the proposed antennas.

When a CLEC is the owner of the structure, the proposed user shall pay the owner directly the one-time charge as set forth above. When the Company is the owner of the support structure, it shall determine the charge on an individual case basis. In the event that a CLEC who owns the support structure fails to comply with these provisions, at the Company's option, ownership of the support structure shall transfer to the Company.

Costs incurred by he Company to conduct a review for wind or ice loadings (etc.) for antennas over 18 inches in diameter, or for any multiple antenna installation, and any changes which may be required thereto in order to insure that such loadings meet generally accepted engineering criteria for radio tower structures, will be billed to the CLEC.

(G) Emergency Power and/or Environmental Support

In the event special work must be done by the Company to provide emergency power or environmental support to the transmitter/receiver equipment or antenna, the CLEC will be billed on a time and materials basis for the costs incurred.

(H) Escorting

When a CLEC is escorted by a qualified Company employee for access to the roof space, transmitter/receiver space, or cable risers and racking for maintenance, the miscellaneous labor charges as set forth in Section 19.16.51 will apply.

(N)

EFFECTIVE: July 12, 2001

Original Page 26,12

FACILITIES FOR INTRASTATE ACCESS

19. COLLOCATION SERVICE

19.14 Rate Regulations (M)

19.14.1 Rates and Charges

(T)

(T)

Except as otherwise described herein, the rates for Company collocation services provided pursuant to this Tariff are set forth in Section 19.16. The tariffed rates herein may be superseded by rates contained in future regulatory orders or as otherwise required by legal requirements.

(Τ)

19.14.2 Billing and Payment

(T)

The initial payment of nonrecurring charges (NRCs) shall be due and payable in accordance with Section 19.4.1. The balance of the NRCs and all related monthly recurring service charges will be billed to the CLEC when the Company provides CLEC access to the caged, cageless or adjacent collocation arrangement and shall be payable in accordance with applicable established payment deadlines.

19.15 Description and Application of Rate Elements

(T)

19.15.1 Non-Recurring Charges

(T)

The following are nonrecurring charges (one-time charges) that apply for specific work activity.

(A) Engineering/Major Augment Fee

(T) (H) (H)

The Engineering/Major Augment Fee applies for each initial caged, cageless, virtual, or microwave collocation request and major augment requests for existing caged, cageless, and virtual collocation arrangements. This charge recovers the costs of the initial walkthrough to determine if there is sufficient collocation space, the best location for the collocation area, what building modifications are necessary to provide collocation, and if sufficient DC power facilities exist in the premises to accommodate collocation. This fee also includes the total time for the Building Services Engineer and the time for the Outside Plant and Central Office Engineers to attend status meetings.

(M)

(M) Material transferred from Page 26.

(N)

Second Revised Page 27 Canceling First Revised Page 27

19. COLLOCATION SERVICE

19.15 Description and Application of Rate Elements (Continued)

19.15.1 Non-Recurring Charges (Continued)

(B) Minor Augment Fee

The Minor Augment Fee applies for each minor augment request of an existing caged, cageless, virtual, or microwave collocation arrangement that does not require additional AC or DC power systems, HVAC system upgrades, or additional cage space. Minor augments are those requests that require the Company to perform a service or function on behalf of the CLEC including, but not limited to: installation of Virtual equipment cards or software upgrades, removal of Virtual equipment, requests to pull cable from exterior microwave facilities and requests to terminate DSO, DS1 and DS3 cables.

(C)

(C) Access Card Administration

The Access Card Administration rate covers activities associated with the issuance and management of premises access cards. The rate is applied on a per card basis.

(D) Cage Enclosure

The Cage Enclosure rate is applied per caged arrangement. This rate includes the labor and materials to recover the costs incurred in constructing the CLEC's cage, cage gate, and grounding bar. There are five caged enclosure rate elements based on the size of the cage: 25 to 100 square feet; 101 to 200 square feet; 201 to 300 square feet; 301 to 400 square feet; and 401 to 500 square feet.

(E) Cage Enclosure Augment

The Cage Enclosure Augment rate is applied per square foot of fencing when a CLEC requests additional fencing for an existing caged arrangement.

(F) BITS Timing

The non-recurring charge for BITS Timing includes engineering, materials, and labor costs to wire a BITS port to the CLEC's equipment. If requested, it is applied on a per project basis.

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA EFFECTIVE: October 13, 2001 ISSUED: September 28, 2001

First Revised Page 28 Canceling Original Page 28

19. COLLOCATION SERVICE

19.15 Description and Application of Rate Elements (Continued)

19.15.1 Non-Recurring Charges (Continued)

(G) Overhead Superstructure

The Overhead Superstructure charge is applied for each initial caged and cageless collocation application. The Overhead Superstructure charge is designed to recover the Company's engineering, material, and installation costs for extending dedicated overhead superstructure.

(H) Facility Pull -Engineering

(C)

The Facility Pull -Engineering charge is applied per project to recover the engineering costs of pulling (C) metallic cable or fiber optic patchcord from the collocation cage or relay rack to the Main Distribution Frame block, DSX panel or fiber distribution panel. The change would also apply per project to recover the engineering costs of pulling transmission cable from microwave antennae facilities on the rooftop to the collocation cage or relay rack.

(I) Facility Pull

The Facility Pull charge is applied per cable run and recovers the labor cost of pulling metallic cable or fiber (C) optic patchcord from the collocation cage or relay rack to the Main Distribution Frame block, DSX panel, or fiber distribution panel.

(C)

(J) Cable Termination

(C)

The Cable Termination charge is applied per cable or fiber optic patchcord terminated and is designed to recover the labor cost of terminating transmission cable or fiber optic patchcord from the collocation cage or relay rack to the Main Distribution Frame block, DSX panel, or fiber distribution panel.

(C)(D)

(D)

EFFECTIVE: July 12, 2001

Third Revised Page 29 Canceling Second Revised Page 29

EFFECTIVE: October 1, 2002

ISSUED: September 16, 2002

19. COLLOCATION SERVICE

19.15 Description and Application of Rate Elements (Continued)

19.15.1 Non-Recurring Charges (Continued)

(K) Fiber Cable Pull-Engineering

The Fiber Cable Pull-Engineering charge is applied per project to cover the engineering costs for pulling the CLEC's fiber cable, when necessary, into the Company's central office.

(L) Fiber Cable Pull-Place Innerduct

The Fiber Cable Pull-Place Innerduct charge is applied per linear foot to cover the cost of placing innerduct. Innerduct is the split plastic duct placed from the cable vault to the CLEC's equipment area through which the CLEC's fiber cable is pulled.

(M) Fiber Cable Pull-Labor

This charge is applied per linear foot and covers the labor costs of pulling the CLEC's fiber cable into the Company's central office.

(N) Fiber Cable Pull-Fire Retardant

This charge is associated with the filling of space around cables extending through walls and between floors with a non-flammable material to prevent fire from spreading from one room or floor to another.

(O) Fiber Splice-Engineering

The Fiber Splice-Engineering charge is applied per project and covers the engineering costs for fiber cable splicing projects.

(P) Fiber Splice

The Fiber Splice charge is applied per fiber cable spliced and recovers the labor cost associated with the splicing.

(Q) DC Power

Non-recurring charges for DC Power are applied for each caged, cageless, and virtual collocation (C) application and major DC Power augments to existing arrangements. These charges recover the Company's engineering and installation costs for pulling and terminating DC power cables to the collocation area. For initial applications, each DC Power feed will require two (2) cables. (C)

First Revised Page 30 Canceling Original Page 30

19. COLLOCATION SERVICE

19.15 Description and Application of Rate Elements (Continued)

19.15.1 Non-Recurring Charges (Continued)

(R) Cable Material Charges

(T)

The CLEC has the option of providing its own cable or the Company may, at the CLEC's request, provide the necessary transmission and power cables. If the Company provides these cables, the applicable Cable Material Charge will be charged.

(T)

(S) Adjacent Engineering Fee

(T)

The Adjacent Engineering Fee provides for the initial activities of the Central Office Equipment Engineer, Land & Building Engineer and the Outside Plant Engineer associated with determining the capabilities of providing Adjacent On-Site collocation. The labor charges are for an on-site visit, preliminary investigation of the manhole/conduit systems, wire center and property, and contacting other agencies that could impact the provisioning of adjacent collocation.

(T) Adjacent Fiber Cable Pull-Engineering

(11)

The Adjacent Fiber Cable Pull-Engineering fee provides for engineering associated with pulling the CLEC's fiber cable in an adjacent collocation arrangement. The Adjacent Fiber Cable Pull-Engineering charge includes the time incurred by the Outside Plant Engineer on the project to determine the conduit/ subduct assignment and associated outside plant activity to complete the work.

(U) Adjacent Fiber Cable Pull-Place Innerduct

(T)

This NRC covers the cost for placing innerduct, if required for adjacent collocation, which is the split plastic duct placed from the cable vault to the CLEC's equipment area through which the CLEC's fiber is pulled.

(V) Adjacent Fiber Cable Pull-Labor

(T)

This charge covers the labor costs for pulling CLEC fiber cable for an adjacent collocation arrangement. Refer to Adjacent Fiber Cable Pull-Engineering above.

(W) Adjacent-Cable Fire Retardant

(T)

This charge is associated with the filling of space around cables extending through walls and between floors with a non-flammable material to prevent fire from spreading from one room or floor to another.

(X) Adjacent Metallic Cable Pull-Engineering

 (Π)

This NRC covers the engineering costs of pulling metallic cable for adjacent collocation into the Company wire center. For adjacent collocation, the metallic cable will be spliced in the cable vault to a stubbed connector located on the vertical side of the main distribution frame to provide proper protection for central office equipment.

(Y) Adjacent Metallic Cable Pull-Labor

(T)

This charge covers the labor costs of pulling metallic cable for adjacent collocation into the Company wire center.

EFFECTIVE: July 12, 2001 ISSUED: June 27, 2001

First Revised Page 31 Canceling Original Page 31

(1)

 (Π)

EFFECTIVE: July 12, 2001

ISSUED: June 27, 2001

19. COLLOCATION SERVICE

19.15 <u>Description and Application of Rate Elements</u> (Continued)

19.15.1 Non-Recurring Charges (Continued) (11) Adjacent Cable Splice-Engineering (Z) This charge covers the outside plant engineering costs for cable splice projects associated with an adjacent collocation arrangement. (11) (AA) Adjacent DS1/DS0 Cable Splice-Greater Than 200 Pair This charge is for the labor to splice metallic cables and is based on a per pair spliced. (AB) Adjacent DS1/DS0 Cable Splice-Less Than 200 Pair (11) This charge is for the labor to splice metallic cables and is based on a per pair spliced. (AC)-(11) Adjacent Fiber Cable Splice This charge covers the labor to splice fiber cables and is based on a per fiber spliced. (AD) Adjacent Facility Pull-Engineering (T) This charge covers the engineering cost associated with the interconnection wire (cable) from the main distribution frame connector to a termination block or DSX panel. (AE) Adjacent Facility Pull-Labor (T) This charge covers the labor of running the interconnection wire (cable) from the main distribution frame connector to a termination block or DSX panel. (AF) Adjacent DS0 Cable Termination (Connectorized)/Adjacent DS0 Cable Termination (Unconnectorized) (Π) These charges cover the labor to terminate these types of interconnection wire (cable) for adjacent collocation to the main distribution frame block or DSX panel. (AG) Adjacent DS1 Cable Termination (Connectorized)/Adjacent DS1 Cable Termination (Unconnectorized) (Π) These charges cover the labor of terminating these types of interconnection wire (cable) for adjacent collocation to the main distribution frame block or DSX panel. (AH) Adjacent DS3 Coaxial Cable Termination (Preconnectorized) /Adjacent DS3 Cable Termination (11) (Unconnectorized) These charges cover the labor of terminating this type of interconnection wire (cable) for adjacent collocation to the main distribution frame block or DSX panel. (AI) Adjacent Fiber Cable Termination (11)

This charge covers the labor of terminating fiber cable, per fiber strand, for adjacent collocation to a

fiber distribution panel.

First Revised Page 32 Canceling Original Page 32

19. COLLOCATION SERVICE

19.15 Description and Application of Rate Elements (Continued)

Non-Recurring Charges (Continued)

(AJ) Collocation Space Report

(T)

When requested by a CLEC, the Company will submit a report that indicates the Company's available collocation space in a particular premises. The report will be issued within ten calendar days of the request. The report will specify the amount of collocation space available at each requested premises, the number of collocators and any modifications in the use of the space since the last report. The report will also include measures that the Company is taking to make additional space available for collocation.

(AK) Miscellaneous Services-Labor

(T)

Additional labor, if required, by the Company to complete a collocation request or perform inventory (C) services for CLECs will be rated as set forth in Section 19.16.

(C)

(AL) Engineering/Major Augment Fee (Microwave Only)

(N)

The Engineering/Major Augment Fee for Microwave Collocation applies when an existing Caged and Cageless collocation arrangement is augmented with newly installed microwave antennae and other exterior facilities. This charge recovers the costs of the initial walkthrough to determine if there is sufficient space, the best location for the microwave antennae and other exterior facilities, what building modifications are necessary, if any, and if sufficient support facilities exist in the premises to accommodate the microwave antennae and other exterior facilities. This fee also includes the total time for the Building Services Engineer to coordinate the entire project.

(AM) Facility Pull (Microwave Only)

The Facility Pull charge is applied per linear foot and recovers the labor cost of pulling transmission cable from the microwave antennae and other exterior facilities on the rooftop to the transmission equipment in the collocation cage or relay rack.

(AN) **Building Penetration for Microwave Cable**

The reasonable costs to penetrate buildings for microwave cable to connect microwave antennae facilities and other exterior facilities to the transmission equipment in the collocation cage or relay rack will be determined and applied on an individual case basis, where technically feasible, as determined by the initial and subsequent Engineering surveys.

(AO)Special Work for Microwave

The costs incurred by the Company for installation of CLEC's microwave antennae and other exterior facilities that are not recovered via other microwave rate elements will be determined and applied on an individual case basis.

(AP) Virtual Equipment Installation

The Virtual Equipment Installation charge is applied on a per quarter rack (or quarter bay) basis and recovers the costs incurred by the Company for engineering and installation of the virtual collocation equipment. This charge would apply to the installation of powered equipment including, but not limited to, ATM, DSLAM, frame relay, routers, OC3, OC12, OC24, OC48, and NGDLC. This charge does not apply for the installation of splitters.

(N)

(M) Material transferred to Page 32.1.

(M) (N)

First Revised Page 32.1 Canceling Original Page 32.1

19. COLLOCATION SERVICE

19.15 <u>Description and Application of Rate Elements</u> (Continued)

19.15.1 Non-Recurring Charges (Continued)

(AQ) Virtual Software Upgrade

The Virtual Software Upgrade charge is applied per base unit when the Company, upon CLEC request, installs software to upgrade equipment for an existing Virtual Collocation arrangement.

(AR) <u>Virtual Card Installation</u>

The Virtual Card Installation charge is applied per card when the Company, upon CLEC request, installs additional cards for an existing Virtual Collocation arrangement.

(AS) <u>Cage Ground Bar</u>

If a CLEC elects to subcontract collocation cage construction to a Company approved contractor, the Company will provision a ground bar in the CLEC's cage. The charge is applied per ground bar and recovers the material and labor costs to install the grounding bar, including necessary grounding wire.

(AT) <u>Dedicated Transit Service (DTS) Service Order Charge</u>

Applied per DTS order to the requesting CLEC for recovery of DTS order placement and issuance costs. The manual charge applies when the semi-mechanized ordering interface is not used.

(AU) <u>Dedicated Transit Service (DTS) – Service Connection CO Wiring</u>

Applied per DTS jumper to the requesting CLEC for recovery of DTS jumper material, wiring, and service turn-up for DS0, DS1, DS3, and dark fiber circuits.

(AV) <u>Dedicated Transit Service (DTS) – Service Connection Provisioning</u>

Applied per DTS order to the requesting CLEC for recovery of circuit design and labor costs associated with the provisioning of DS0, DS1,DS3 and dark fiber circuits for DTS.

(M)

(N)

(N)

(N)

19. COLLOCATION SERVICE

19.15 <u>Description and Application of Rate Elements</u> (Continued)

(N)

19.15.2 Monthly Charges

(M)

The following are monthly charges. Monthly charges apply each month or fraction thereof that Collocation Service is provided.

(A) Caged Floor Space

Caged Floor Space is the cost per square foot to provide environmentally conditioned caged floor space to the CLEC. Environmentally conditioned space is that which has proper humidification and temperature controls to house telecommunications equipment. The cost includes only that which relates directly to the land and building space itself.

(B) Relay Rack Floor Space

The Relay Rack Floor Space charge provides for the environmentally conditioned floor space that a relay rack occupies based on linear feet. The standardized relay rack floor space depth is based on half the aisle area in front and back of the rack, and the depth of the equipment that will be placed within the rack.

(C) Cabinet Floor Space

The Cabinet Floor Space charge provides for the environmentally conditioned floor space that a telecommunications equipment cabinet occupies based on linear feet. The standardized floor space depth is based on the size of the cabinet and half of the aisle in the front and rear of the cabinet. The cabinet size is based on the Company's standard cabinet size of 33 inches by 29 inches.

(D) Cable Subduct Space-Manhole

This charge applies per project per month and covers the cost of the space that the outside plant fiber occupies within the manhole.

(E) Cable Subduct Space

The Subduct Space charge covers the cost of the subduct space that the outside plant fiber occupies and applies on a per linear foot basis.

(M)

EFFECTIVE: October 13, 2001

ISSUED: September 28, 2001

Third Revised Page 33 Canceling Second Revised Page 33

(C)

19. COLLOCATION SERVICE

19.15 Description and Application of Rate Elements (Continued)

19.15.2 Monthly Charges (Continued)

(F) Fiber Cable Vault Splice

The Fiber Cable Vault Splice charge applies per subduct or per splice and covers the space and material cost associated with the CLEC's fiber cable splice within the Company's cable vault.

(G) Cable Rack Space-Metallic

The Cable Space-Metallic charge is applied for each DS0, DS1 and DS3 cable run. The charge is designed to recover the space utilization cost that the CLEC's metallic and coaxial cable occupies within the Company.

(H) Cable Rack Space-Fiber

The Cable Rack Space-Fiber charge recovers the space utilization cost that the CLEC's fiber cable occupies within the Company's cable rack system.

(1) DC Power

The DC Power monthly charge is applied on a per load amp basis with a 10 amp minimum for each (C) caged, cageless, and virtual collocation arrangement. This charge is designed to recover the monthly facility and utility expense to power the collocation equipment.

(J) **Facility Termination**

This charge is applied per cable terminated. This charge is designed to recover the labor and material costs of the applicable main distribution frame 100 pair circuit block, DSX facility termination panel, or fiber distribution panel.

(K) **BITS Timing**

The BITS Timing monthly charge is designed to recover equipment and installation cost to provide synchronized timing for electronic communications equipment. This rate is based on a per port cost.

(L) **Building Modification**

The Building Modification monthly charge is applied to each caged and cageless arrangement and is associated with provisioning the following items in the Company's premises; security, dust partition, ventilation ducts, demolition/site work, lighting, outlets, and grounding equipment.

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

EFFECTIVE: October 1, 2002 ISSUED: September 16, 2002

Third Revised Page 34
Canceling Second Revised Page 34

19. COLLOCATION SERVICE

19.15 Description and Application of Rate Elements Continued)

19.15.2 Monthly Charges (Continued)

(M) Environmental Conditioning

The Environmental Conditioning charge is applied to each caged, cageless, and virtual arrangement on a per load (C) amp increment (10 amp minimum) based on the CLEC's DC Power requirements. This charge is associated with (C) the provisioning of heating, ventilation, and air conditioning systems for the CLEC's equipment in the Company's premises.

(N) Adjacent Cable Subduct Space-Manhole

This charge covers the space utilization cost that the outside plant fiber or metallic cable occupies within the manhole.

(O) Adjacent Cable Subduct Space

The Adjacent Cable Subduct Space charge covers the space utilization cost of the subduct that the outside plant fiber or metallic cable occupies within the conduit system.

(P) Adjacent Conduit Space (Metallic)-Manhole

This charge covers the space utilization cost that the outside plant metallic cable occupies within the manhole.

(Q) Adjacent Conduit Space (Metallic)

This charge covers the space utilization cost that the outside plant metallic cable occupies within the conduit system.

(R) Adjacent Facility Termination DS0 Cable

This charge is applied per 100 pair cable terminated. This charge is designed to recover the labor and material cost of the main distribution frame 100 pair circuit block.

(S) Adjacent Facility Termination DS1 Cable

The Facility Termination (DS1) charge is applied per 28 pair DS1 cable terminated. This charge is designed to recover the labor and material cost of the DSX facility termination panel.

(T) Adjacent Facility Termination DS3 Cable

The Facility Termination (DS3) charge is applied per DS3 cable terminated. This charge recovers the labor and material cost of the DSX facility termination panel.

(U) Adjacent Cable Vault Space

The Adjacent Cable Vault Space charge covers the cost of the space the CLEC's cable occupies within the cable vault. The charge is based on the diameter of the cable or subduct.

(V) Adjacent Cable Rack Space

This charge covers the space utilization cost that the CLEC's fiber, metallic or coaxial cable occupies within the cable rack system. The charge is based on the linear feet occupied.

(W) Microwave Rooftop Space

Microwave Rooftop Space is the cost per square foot to provide rooftop space to the CLEC for microwave antennae and other exterior facilities. The cost includes only that which relates directly to the land and building space itself.

(X) Virtual Equipment Maintenance

The Virtual Equipment Maintenance charge is applied on a per quarter rack (or quarter bay) basis and recovers the costs incurred by the Company for maintenance of the CLEC's virtual collocation equipment. This charge would apply to the maintenance of equipment including, but not limited to: ATM, DSLAM, frame relay, routers, OC3, OC12, OC24, OC48, and NGDLC. This charge does not apply for the maintenance of splitters.

EFFECTIVE: October 1, 2002

ISSUED: September 16, 2002

First Revised Page 35 Canceling Original Page 35

19. COLLOCATION SERVICE

19.16	Rates and Charges		
		NRC	MONTHLY CHARGE
		<u> </u>	OHAROL
19.16.1	Engineering/Major Augment Fee, Per Occurrence	# 4400.00	(T)
	Caged or Cageless Virtual	\$ 1,129.00 972.80	(T) (N)
	Microwave Augment	1,741.02	(N) (N)
		η, π.σ.	(1)
19.16.2	Minor Augment Fee, Per Occurrence	200.00	
19.16.3	Access Card Administration, Per Card	21.00	
19.16.4	Cage Enclosure		
	25-100 Sq.Ft. Cage	4,907.00	
	101-200 Sq.Ft. Cage	5,661.00	-
	201-300 Sq.Ft. Cage	7,021.00	-
	301-400 Sq.Ft. Cage	8,382.00	
	401-500 Sq.Ft. Cage	9,742.00	
19.16.5	Cage Enclosure Augment		
	Per Sq. Ft. Fencing	13.00	
19.16.6	BITS Timing,		
	Per Project	307.00	_
	Per Month		\$11.00
19.16.7	Overhead Superstructure, Per Project	2,372.00	
19.16.8	Facility Pull- Engineering		
	Per Project	72.00	-
19.16.9	Facility Pull, Per Cable Run		
	Metallic	190.00	– (C)
	Fiber Optic Patchcord	172.05	(N)
19.16.10	Cable Termination		
	DSO Cable, per 100 pair cable		
	terminated	4.00	-
-	DS1 Cable, per 28 pair cable		
	terminated	1.00	
	DS3 Coaxial Cable (Preconnectorized), per cable		(T)
	terminated	1.00	
	DS3 Coaxial Cable (Unconnectorized), per cable	40.00	(11)
	terminated	10.00	
	Fiber Optic Patchcord, per termination	.93	(N)
19.16.11	Fiber Cable Pull - Engineering		
	Per Project	607.00	
19.16.12	Fiber Cable Pull – Place Innerduct		
	Per Linear Foot	2.00	
10 16 12	Fiber Cable Pull - Labor		
	Per Linear Foot	1.00	_

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

Third Revised Page 36 Canceling Second Revised Page 36

19. COLLOCATION SERVICE

19.16	Rates and Charges (Continued)		
		1150	MONTHLY
40.40.4	L Eiber Cable Dull Cable - Eiro Deterdant	NRC	<u>CHARGE</u>
19.16.14	Fiber Cable Pull-Cable - Fire Retardant Per Occurrence	\$ 38.00	_
		,	
19.16.15	Fiber Splice – Engineering		
	Per Project	31.00	
19 16 16	Fiber Cable Splice		
	Per Fiber	45.00	
19.16.17	DC Power		
	Engineering, Per Project	71.44	(1)
	Cable Pull/Termination, Per Cable	1,213.86	(C)
	Ground Wire, Per Wire	18.12	- (N)
	Facility and Utility, Per Load Amp (10 Amp Min)	-	\$15.00 (N)
19 16 18	Cable Material Charge		
10110110	Facility Cable, Per Cable Run		
	DSO Cable (Connectorized)		
	100 Pair	331.00	
	DS1 Cable (Connectorized)	307.00	
	DS3 Coaxial Cable	84.00	
		34.00	
	Shielded Cable (Orange Jacket)		-
	Fiber Optic Patchcord-24 Fiber (Connectorized)	836.20	-
	Facility Cable, Per linear Foot	4.40	
	Category 5 (Connectorized)	1.10	
	Power Cable, Per Cable Run		
	Wire Power 1/0	93.00	
	Wire Power 2/0	135.00	
	Wire Power 3/0	149.00	
	Wire Power 4/0	184.00	
	Wire Power 350 MCM	313.00	
	Wire Power 500 MCM	437.00	
	Wire Power 750 MCM	673.00	-
19.16.19	Caged Floor Space, Per Square Foot,		
	Per Month	***	3.00
19.16.20	Relay Rack Floor Space,		40.00
	Per Linear Foot, Per Month		12.00
19.16.21	Cabinet Floor Space, Per Linear Foot,		
	Per Month		16.00
10 16 22	Cable Subduct Space-Manhole		
13.10.22	Per Project, Per Month		6.00
	rei rioject, ret ivioliut		0.00

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA EFFECTIVE: October 1, 2002 ISSUED: September 16, 2002

Third Revised Page 37 Canceling Second Revised Page 37

19. COLLOCATION SERVICE

19.16	Rates and Charges (Continued)		
		NRC	MONTHLY <u>CHARGE</u>
19.16.23	Cable Subduct Space		OHANGE
	Per Linear Foot, Per month	-	\$.04
19 16 24	Fiber Cable Vault Splice-48 Fiber		
13.10.24	Material, Per Splice, Per month		10.00
19.16.25	Fiber Cable Vault Splice-48 Fiber		
	Per Subduct, Per month		1.00
19.16.26	Fiber Cable Vault Splice-96 Fiber		
	Material, Per Splice, Per Month	_	27.00
19.16.27	Fiber Cable Vault Splice-96 Fiber		
	Per Subduct, Per Month –		1.00
19.16.28	Cable Rack Space-Metallic		
	Per Cable Run, Per Month		2.00
19.16.29	Cable Rack Space-Fiber		
	Fiber Entrance Facility Cable, Per Innerduct Foot, Per Month		.02
	Fiber Optic Patchcord to Fiber Distribution Frame, Per Cable, Per Month	-	.55
19.16.30	Facility Termination		
	DSO, Per 100 Pair, Per Month		4.00
	DS1, Per 28 Pair, Per Month		16.00
	DS3, Per DS3, Per Month	-	11.00
	Fiber Optic Patchcord, Per Connector, Per Month		1.07
19.16.31	Building Modification		
	Per Project		183.00
19.16.32	Environmental Conditioning		
	Per Load Amp DC Power (10 Amp Min)		3.00 (T)
19.16.33	Adjacent Engineering Fee		
	On-Site, Per Occurrence	\$958.00	

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

EFFECTIVE: October 1, 2002 ISSUED: September 16, 2002 VERIZON FLORIDA INC.

FACILITIES FOR INTRASTATE ACCESS

First Revised Page 38 Canceling Original Page 38

19. COLLOCATION SERVICE

19.16	Rates and Charges (Continued)			(T)
19.16.34	Adjacent Fiber Cable Pull- Engineering, Per Project	NRC \$ 607.00	MONTHLY CHARGE 	(T)
19.16.35	Adjacent Fiber Cable Pull-Place Innerduct, Per Linear Foot	2.00	**	(T)
19.16.36	Adjacent Fiber Cable Pull- Per Linear Foot	1.00		(T)
19.16.37	Adjacent Cable Fire Retardant- Per Occurrence	38.00	-	(T)
19.16.38	Adjacent Metallic Cable Pull- Engineering, Per Project	607.00		(T)
19.16.39	Adjacent Metallic Cable Pull Per Linear Foot	1.00	_	(T)
19.16.40	Adjacent Metallic Cable Splice Engineering, Per Project Greater than 200 Pair,	31.00		(T)
	Per DSO/DS1 Pair 200 Pair or Less, Per DSO/DS1 Pair	1.00 2.00	-	(T)
19.16.41	Adjacent Fiber Cable Splice Engineering, Per Fiber 48 Fiber or Less, Per Fiber Greater than 48 Fiber, Per Fiber	31.00 45.00 40.00	 - 	(T)
19.16.42	Adjacent Facility Pull-Engineering Per Project	72.00		(T)

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

First Revised Page 39 Canceling Original Page 39

EFFECTIVE: July 12, 2001 ISSUED: June 27, 2001

19. COLLOCATION SERVICE

19.16	Rates and Charges (Continued)			(T)
		NRC	MONTHLY CHARGE	
19.16.43	Adjacent Facility Pull			(T)
	Per Linear Foot	1.00	-	
1916.44	Adjacent Cable Termination			(T)
	DSO Cable (Connectorized)			
	Per 100 Pair	4.00		
	DSO Cable (Unconnectorized)	•		
	Per 100 Pair	38.00		
	DS1 Cable (Conncetorized)			
	Per 28 Pair	1.00		
	DS1 Cable (Unconnectorized)			
	Per 28 Pair	29.00	-	
	DS3 Coaxial Cable (Preconnectorized)			
	Per DS3	1.00		
	DS3 Coaxial Cable (Unconnectorized)			
	Per DS3	10.00		
	Fiber Cable, Per Fiber Termination	45.00		
19.16.45	Adjacent Subduct Space			(T)
	Manhole, Per Project		6.00	` '
	Per Linear Foot	-	.04	
19.16.46	Adjacent Conduit Space (4" Duct)			(T)
	Manhole, Metallic, Per Conduit	_	10.00	` '
	Metallic, Per Linear Foot	_	.04	
19.16.47	Adjacent Facility Termination-Material			(T)
	DSO Cable, Per 100 Pair		4.00	. ,
	DS1 Cable, Per 28 Pair		16.00	
	DS3 Cable, Per Coaxial		11.00	
	• • • • • • • • • • • • • • • • • • • •			

First Revised Page 40 Canceling Original Page 40

19. COLLOCATION SERVICE

19.16 Rates and Charges (Continued)

	NRC	MONTHLY CHARGE	
19.16.48 Adjacent Cable Vault Space		¢404.00	
Per 1200 Pair, Material, Per Splice		\$461.00 4.00	
Per 1200 Pair, Per Cable		337.00	
Per 900 Pair, Material, Per Splice Per 900 Pair, Per Cable		3.00	
Per 600 Pair, Material, Per Splice	***	225.00	
Per 600 Pair, Per Cable		3.00	
Per 100 Pair, Material, Per Splice		47.00	
Per 100 Pair, Per Cable		1.00	
Per 48 Fiber, Material, Per Splice	-	10.00	
Per 48 Fiber, Per Subduct	-	1.00	
Per 96 Fiber, Material, Per Splice		27.00	
Per 96 Fiber, Per Subduct	-	1.00	
19.16.49 Adjacent Cable Rack Space		0.4	
Metallic DSO, Per Linear Foot		.01	
Metallic DS1, Per Linear Foot		.01	
Fiber, Per Innerduct Foot	-	.02 .01	
Coaxial, Per Linear Foot		.01	
19.16.50 Collocation Space Report-Per Wire			
Center or Access Tandem Requested	\$1,218.00		
19.16.51 Labor Rates, Per Technician Basic Business Day			
1록 Half Hour	41.66		
Each Additional Half Hour	20.83		
Overtime Non-Business Day			
1 st Half Hour	100.00		
Each Additional Half Hour	75.00		
Premium Non-Business Day	450.00		
1st Half Hour	150.00 125.00	_	
Each Additional Half Hour	125.00	-	
19.16.52 Facility Pull (Microwave Only), Per Linear Foot	.93	- (N	l 1)
19.16.53 Microwave Rooftop Space Per Square Foot, Per Month	-	2.76	
19.16.54 Building Penetration for Microwave Cable, Per Occurrence	ICB		
19.16.55 Special Work for Microwave, Per Occurrence	ICB		
19.16.56 Virtual Equipment Installation, Per Quarter Rack	3,440.64	-	
19.16.57 Virtual Software Upgrade, Per Base Unit	79.45	-	
19.16.58 Virtual Card Installation, Per Card	223.62	-	
19.16.59 Virtual Equipment Maintenance, Per Quarter Rack		71.47	
19.16.60 Cage Ground Bar	1,270.93	- (1	N)

JOHN P. BLANCHARD, PRESIDENT TAMPA, FLORIDA

19. COLLOCATION SERVICE

19.16	Rates and Charges (Continued)		(N)
		NRC	MONTHLY CHARGE
19.16.61	Dedicated Transit Service (DTS)		
	DSO		
	Service Order, Semi-Mechanized, per order	31.50	
	Service Order, Manual, per order	49.60	
	Service Order CO Wiring, per jumper	6.64	
	Service Connection Provisioning, per order	108.72	
	DS1, DS3, Dark Fiber		
	Service Order, Semi-Mechanized, per order	31.50	
	Service Order, Manual, per order	49.60	
	Service Order CO Wiring, per jumper	16.10	
	Service Connection Provisioning, per order	109.08	(N)